Pepperdine University
Pepperdine Digital Commons

Theses and Dissertations

2018

# A case study of a K-12 learning center in Southern California: exploring strategies to sustain learning centers for students with learning disabilities

Rebecca Michelle Cohen

Follow this and additional works at: https://digitalcommons.pepperdine.edu/etd

### **Recommended Citation**

Cohen, Rebecca Michelle, "A case study of a K-12 learning center in Southern California: exploring strategies to sustain learning centers for students with learning disabilities" (2018). *Theses and Dissertations*. 954.

https://digitalcommons.pepperdine.edu/etd/954

This Dissertation is brought to you for free and open access by Pepperdine Digital Commons. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of Pepperdine Digital Commons. For more information, please contact josias.bartram@pepperdine.edu, anna.speth@pepperdine.edu.

Pepperdine University

Graduate School of Education and Psychology

# A CASE STUDY OF A K-12 LEARNING CENTER IN SOUTHERN CALIFORNIA: EXPLORING STRATEGIES TO SUSTAIN LEARNING CENTERS FOR STUDENTS WITH LEARNING DISABILITIES

A dissertation submitted in partial satisfaction

of the requirements for the degree of

Doctor of Education in Educational Leadership, Administration, and Policy

by

Rebecca Michelle Cohen

June, 2018

Paula Thompson, Ed.D. - Dissertation Chairperson

This dissertation, written by

## Rebecca Michelle Cohen

under the guidance of a Faculty Committee and approved by its members, has been submitted to and accepted by the Graduate Faculty in partial fulfillment of the requirements for the degree of

## DOCTOR OF EDUCATION

Doctoral Committee:

Paula Thompson, Ed.D., Chairperson Kay Davis, Ed.D. Linda Purrington, Ed.D.

© Copyright by Rebecca Cohen (2018) All Rights Reserved

# TABLE OF CONTENTS

LIST OF TABLES	vii
LIST OF FIGURES	viii
DEDICATION	ix
VITA	x
ABSTRACT	xii
Chapter One: The Problem	1
Background of the Study	1
Problem Statement	4
Purpose	4
Importance of the Study	4
Definitions of Terms	6
Theoretical Framework (Summary)	8
Research Questions	9
Limitations	
Delimitations	
Assumptions	
Organization of the Study	
Chapter Two: Literature Review	13
Theoretical Framework	14
Historical Context	
Literature Themes	
Summary	
Chapter Three: Methodology	
Purpose	
Research Questions	
Research Design and Rationale	
Setting	
Population, Sample, and Sampling Procedures	

Human Subject Considerations	
Instrumentation	
Data Collection Procedures	
Data Management	64
Data Analysis	
Positionality	
Chapter Four: Report of the Findings	
Purpose	
Research Questions	
Interviewee Demographics	
Artifact Collection	
Observation	71
Data Summary	
Research Sub-questions	
Themes	
Summary of Key Findings and Themes	
	0.2
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement	
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement	
Chapter Five: Conclusion, Implications, and Recommendations	
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement Purpose	
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement Purpose Theoretical Framework (Summary)	
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement Purpose Theoretical Framework (Summary) Research Questions	
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement Purpose Theoretical Framework (Summary) Research Questions Discussion of Key Findings	
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement Purpose Theoretical Framework (Summary) Research Questions Discussion of Key Findings Implications for Practice	
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement Purpose Theoretical Framework (Summary) Research Questions Discussion of Key Findings Implications for Practice Conclusions	
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement Purpose Theoretical Framework (Summary) Research Questions Discussion of Key Findings Implications for Practice Conclusions Recommendations for Further Study	
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement Purpose Theoretical Framework (Summary) Research Questions Discussion of Key Findings Implications for Practice Conclusions Recommendations for Further Study Concluding Thoughts	93 93 94 94 94 94 95 100 100 100 106 107 111
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement Purpose Theoretical Framework (Summary) Research Questions Discussion of Key Findings Implications for Practice Conclusions Recommendations for Further Study Concluding Thoughts REFERENCES	93 93 94 94 94 94 95 100 100 100 106 107 111 111 121
Chapter Five: Conclusion, Implications, and Recommendations Problem Statement Purpose Theoretical Framework (Summary) Research Questions Discussion of Key Findings Implications for Practice Conclusions Recommendations for Further Study Concluding Thoughts REFERENCES APPENDIX A: Recruitment Form	

v

APPENDIX E: Interview Protocol and Questions for Owners/Managers	126
APPENDIX F: Interview Protocol and Questions for Employees	128
APPENDIX G: Archival Data Collection Guide	129

# LIST OF TABLES

	Page
Table 1. The Framework for Business Engagement in Education With the Overlapping         Literature Themes.	17
Table 2. The Resource Based Theory of Entrepreneurship With the Overlapping Literature Themes.	22
Table 3. Relationships among Guiding Questions, Interview Questions, and Literature	59

Table 3. Relationships among Guiding Questions, Interview Questions, and Literature	59
Table 4. Relationships among Guiding Questions, Archival Data Collection Categories, and Literature.	
Table 5. Interviewee Information	69
Table 6. Artifact Information	70
Table 7. Code Groups and Codes.	73
Table 8. Data Word Count: Top 25 Descriptive Words Used	74

# LIST OF FIGURES

	Page
Figure 1. Hand-written observation notes	72
Figure 2. Word cloud: Top 25 descriptive words used	75

### DEDICATION

This dissertation is dedicated to my husband, Yermie, who has stood by me and supported me through my entire post-graduate journey into the world of education. Thank you for all of the advice, love, and humor you shared each day to propel me to be the person I am today.

### VITA

### **REBECCA COHEN**

### **EDUCATION**

**Doctorate of Education: Educational Leadership, Administration, and Policy,** Expected May 2018, Pepperdine University, Los Angeles, CA, Graduate School of Education and Psychology

**Master of Arts in Education: Educational Psychology,** December 2014, Pepperdine University, Los Angeles, CA, Graduate School of Education and Psychology

**Bachelor of Arts in Theater Arts: Emphasis in Stage Management,** May 2010, Pepperdine University, Malibu, CA, Seaver College

### **PROFESSIONAL EXPERIENCE**

### Founder and Educational Therapist

The Learning Boulevard, Los Angeles, CA, December 2015 - Present

- Provide educational therapy to students one-on-one in home and at the learning center
- Individualize instruction for each student to cater to their specific learning styles
- Work with students academically, socially, and behaviorally to achieve positive results

### **Private Instructor and Learning Specialist**

Los Angeles, CA, June 2014 – December 2015

- Individualized instruction to elementary and middle school students
- Incorporation of the latest educational technologies and practices
- Communication with families of students including reports, assessments, and meetings

### **Director of Day School Programs**

### **Reading Specialist and Homeschool Teacher**

Groza Learning Center, Pacific Palisades, CA, March 2015 – November 2015

- Creation and implementation of individualized curricula for students in K-8
- Management of teachers and collaboration through planning and projects
- Incorporation of online learning applications in the classroom
- One-to-one teaching students to state standards and overseeing classroom management
- Execution of reading remediation programs

• Administration and analysis of a variety of reading diagnostics

# **Behavior** Therapist

# Inclusive Education and Community Partnership (IECP), Los Angeles, CA,

August 2014 – March 2015

- Provided school-based behavioral services to individuals with disabilities
- Implemented ABA intervention plans and recorded student progress
- Collaborated with students, parents, and educators to create an inclusive environment

# Assistant to the Performing Arts Department Assistant to the Business Office

Brentwood School, Los Angeles, CA, August 2012 – May 2014

- Mentored various performing arts students and assisted in classrooms
- Collaborated on projects and shows with parents, students, and educators
- House Manager for each performance and management of ticketing website and publicity
- Assisted departments by processing all paperwork and purchases, collaborating in meetings, and cataloging

# ADDITIONAL RELEVANT EXPERIENCE

- Educational Program Proficiencies: Lindamood Bell; Phonetics; Whole Language; Common Core
- Diagnostic Testing Instruction: GORT, GSERT, SORT, LAC3, Symbol Imagery, Phonemic Awareness, DIBELS
- Intervention Nonviolent Crisis Intervention (NCI) Trainer for Crisis Prevention Institute (CPI)
- CBEST: Passed on 10/04/2014
- Computer Proficiencies: Blackboard, Blackbaud, Schoology, Aleks Math, MS Office, PowerPoint, HTML, Adobe Photoshop, Adobe InDesign, iMovie

### ABSTRACT

The varied academic needs of students with learning disabilities throughout the U.S. and in Southern California, specifically, have driven demand for private learning centers. For the purposes of this study, a learning center refers to a private business that teaches primary and secondary school students with learning disabilities outside of the school system. However, these centers often struggle with business success and the retention of employees and clients. Little research exists to address this topic. Therefore, there is a need to explore strategies to sustain these centers for students with learning disabilities. The purpose of this qualitative case study was to explore strategies used by a successful K-12 learning center for students with learning disabilities in Southern California to sustain their business. Data was collected from interviews with four employees, observations from four visits, and seven types of archival data. The data were analyzed and grouped into the five deductive themes found in the literature review: training, relationships, innovation, structure, and customer development. Five new inductively developed themes resulted from the analysis of the data: Theme 1: Engaging in closed-loop communication with all stakeholders; Theme 2: Taking a holistic approach to student improvement; Theme 3: Providing differentiated instruction for a personalized experience; Theme 4: Engaging in a growth mindset; Theme 5: Setting an intention for the learning center to follow. Aspects such as innovation, flexibility, and intentionality proved to be beneficial to improving student outcomes and sustaining a learning center. Three conclusions were made from the study: Conclusion 1: A learning center can be sustained through a focus on the five literature themes of training, relationships, innovation, structure, and customer development; Conclusion 2: Flexibility allows for individualization, and continual improvement; Conclusion 3: Learning centers seeking a competitive advantage should focus on innovation.

### **Chapter One: The Problem**

#### **Background of the Study**

In the United States, students are continually placed under pressure to fulfill academic requirements and meet state standards (Segool, Carlson, Goforth, von der Embse, & Barterian, 2013). The standards, labeled the Common Core State Standards (CCSS), are detailed, academic standards that relate to the real world and are meant to give students the necessary skills and knowledge to succeed in future schooling and careers. The CCSS provide detailed descriptions for parents and teachers so that they can better assist the students (Caruana, 2015). These standards present an especially monumental challenge for students with learning disabilities (Caruana, 2015; Haager & Vaughn, 2013). The current methods to help these students in the school system involve Individualized Education Programs (IEPs), resource rooms with special educators, and differentiated instruction (Caruana, 2015; Haley, Hammond, Ingalls, & Marín, 2013; Sadioglu, Bilgin, Batu, & Oksal, 2013; Stroggilos & Xanthacou, 2006; Zickel & Arnold, 2001).

Unfortunately, many students with learning disabilities, especially those in public schools, are not always able to get the attention they need during the school day and are in need of additional support (Haager & Vaughn, 2013; Haley et al., 2013; Sadioglu et al., 2013). Even though the IEP has been included in education since 1975, it was not until the Individuals with Disabilities Education Improvement Act (IDEA) in 2004 that more of a focus was placed on students with disabilities. IDEA requires public schools to have an IEP for every student with disabilities who meet the special education requirements for federal and state standards. Since IDEA, IEPs contain goals that are based on state standards, some of which require an overwhelming amount of progress for students to meet. Due to IDEA, more teachers are asking

parents to seek support and accommodations (Caruana, 2015). Teachers recommend that their students are aided by academic professionals beyond the school environment (Haager & Vaughn, 2013; Haley et al., 2013; Sadioglu et al., 2013). As a result of this growing need, learning centers, providing professional support and accommodations for students with learning disabilities outside of school hours, have become an integral part of the education system.

Learning center is a broad term and can take on many meanings with regards to primary and secondary school students, college students, and even those in assisted living situations (Stahl & Henk, 1993). Learning centers are a place where individualized learning is available for students based on their needs (Christ, 1971). Current research on learning centers is limited and is generally based on higher education facilities attached to universities (Norton, 2002; Payne, Hodges, & Hernandez, 2017; Truschel & Reedy, 2009, Truuvert, 2014). For the purposes of this study, a learning center refers to a private business that teaches primary and secondary school students with learning disabilities outside of the school system. As an example of the growth of learning centers over the 21st century, Park, Buchmann, Choi, and Merry (2016) calculated how many fifteen-year-old students participated in academic assistance outside of school in the years 2003 and 2012 in various countries. In the nine-year span, the amount of fifteen-year-old students in supplementary education rose significantly in each country. In the United States, the percentage of these students in tutoring rose from 11.7% to 17%. This shows the growth of students participating in tutoring as well as the growth of accessibility for supplementary education centers. While plenty of learning centers for students with disabilities are available, these centers are created by educators who may not have the entrepreneurial or business mindset needed to sustain and grow their centers in order to give their students opportunities to succeed.

In relating business to education, schools are run both as educational institutions and as businesses to keep their facilities updated and their staff employed (Baker, Campbell, & Ostroff, 2016; Levine & Levine, 2014). When it comes to the three main types of schools, public, private, and charter, a bountiful amount of research exists on what is needed from a business perspective to keep these schools running. Public schools receive money from the government through the help of tax dollars and allow children to attend for free (Cornman, Zhou, Howell, & Young, 2017). Private, or independent schools receive money from tuition, grants, donations, and endowments and are not free to the families whose students attend (Baker et al., 2016). Charter schools are a hybrid of public and private schools. They are technically public schools and children may attend for free, but there is limited space. Charter schools receive money from the government and are sometimes associated with for-profit organizations (Levine & Levine, 2014). The business aspects that each of these schools have in common are that there are a team of staff hired to attend to business management and monetary allocation. This allows these institutions to educate their students and care for their employees (Baker et al., 2016; Cornman et al., 2017; Levine & Levine, 2014). However, this is not the same for learning centers. Learning centers, unlike school models, resemble the model of small businesses (Aurini, 2012; Christ, 1971; Stahl & Henk, 1993).

Though there is currently a lack of research on the success and needs of learning centers from a business perspective, plenty of prior research demonstrates what is needed to operate small businesses in other industries (Gerhardt, Hazen, & Lewis, 2014; Huang & Cho, 2010; Oana, 2012; Omri, Frikha, & Bouraoui, 2015; Philip, 2011; Rese & Baier, 2011; Simpson, Tuck, & Bellamy, 2004; Walker & Brown, 2004; Yang, 2016; Zapalska, Brozik & Zieser, 2015). As learning centers are a type of small business, information learned from other types of small businesses may be relatable. Research on the success of these businesses may provide insight into the needs of the operation of learning centers. However, there is a gap in the literature not only about the success of primary and secondary school student learning centers in general, but also detailing the business strategies of these centers in order to sustain their business.

### **Problem Statement**

Currently, the varied academic needs of students with learning disabilities throughout the U.S. and in Southern California, specifically, has driven demand for private learning centers. When student needs are not fully met within the school system during school hours, teachers recommend and parents seek outside help from learning centers (Park et al., 2016). However, these learning centers often struggle with overall business success and the retention of employees and clients alike. Therefore, there is a need and an opportunity to explore strategies to sustain learning centers for students with learning disabilities through the study of one successful learning center.

### Purpose

The purpose of this qualitative case study was to explore and describe strategies used by a successful K-12 learning center for students with learning disabilities in Southern California to sustain their businesses. This case study took place at one purposefully selected learning center catered to students with learning disabilities in Southern California that was currently successful in their ability to sustain their business. This qualitative case study incorporated data collection through interviews, observations, and archival data (Yin, 2014).

### **Importance of the Study**

This study is an important addition to the educational literature for several reasons: the current lack of research on this topic, the increased need for learning centers for students with

disabilities, and the potential strategies for owners of learning centers to assist in the growth of their businesses. Currently, a lack of empirical research has been conducted on learning centers and similar educational businesses directed toward primary and secondary school students. There is limited data about learning centers, what makes them high-performing, and what strategies they use to sustain their current businesses in order to assist students academically. The most recent study depicting the details of a private learning center for K-12 students is a dissertation discussing an assessment tool for the center that was written over two and a half decades ago (Zoll, 1990).

With the increased need for these centers, educational therapists, tutors, and learning center owners and managers of may benefit from this study by understanding the contributing factors that allow these businesses to sustain and grow. In addition to teachers benefitting, students with learning disabilities may benefit as well through the expansion of centers created to assist them. By exploring successful business strategies for learning centers, current centers may have a better chance of survival and new centers may emerge, fulfilling the need for assistance outside of the classroom. Through the study of centers that are currently successful in their ability to sustain and grow, the findings of the research have the opportunity to provide helpful strategies to learning center businesses. The findings of this study could lead to beneficial strategies for owners of centers to assist in their business growth. It could also lead to the opportunity for education entrepreneurs to create new learning centers, which paves the way of benefiting a larger population of students with learning disabilities.

### **Definitions of Terms**

The following are conceptual and operational definitions of terms as they relate to this study:

- *Customer development*. Customer development is one of the five literature themes used in this study. Customer development is understanding the needs of a client base to better serve the population of a business (Bush, 2016; King-Sears, 2007; Yang, 2016; Zickel & Arnold, 2001).
- *Educational therapist*. Educational therapy refers to the broad practice of activities ranging "from academic tutoring to psychotherapy" (Radecki, 1984, p. 7). An educational therapist focuses on the best learning process for students. They use specific goals and techniques catered to each students' needs, especially involving students with learning disabilities (Radecki, 1984).
- *Entrepreneurship*. Entrepreneurship is seeking out new and unknown opportunities beyond what one currently possesses (Kellermanns, Walter, Crook, Kemmerer & Narayanan, 2016).
- *Innovation.* Innovation is one of the five literature themes used in this study. Innovation is the creation of new ideas, concepts, structures, and resources in order to have a unique business that provides services unparalleled to others (Bond, 2012; Omri et al., 2015; Petersen, 2014; Rese & Baier, 2011).
- *Learning center*. For the purposes of this research study, a learning center is a small business that may or may not be a franchise. Learning centers are independent from school systems and give primary and secondary school students the opportunity to

advance their learning and receive extra academic support (Aurini, 2012; Christ, 1971; Stahl & Henk, 1993).

- *Learning disability*. A learning disability is a condition involving difficulties in gaining knowledge and abilities at the same rate as one's peers. These disabilities are generally not associated with a physical handicap but does not necessarily exclude those with such handicaps (Cornucopia of Disability Information, 2007; Haager & Vaughn, 2013; Radford et al., 2015).
- *Relationships*. Relationships is one of the five literature themes used in this study.
   Relationships refers to networking and making a connection with individuals through varying mediums both digitally and in person (Bolman & Deal, 2017; Epstein, 2002; Jochims, 2016).
- *Resource-based theory (RBT)*. RBT is a business theory with conditions related to the use of diverse resources that are theorized to lead to business success (Alvarez & Busenitz, 2001; Kellermanns et al., 2016).
- *Small business*. For the purposes of this research study, a small business is classified as a company that is independently owned and operated that may or may not be franchised (Simpson, Tuck, & Bellamy, 2004).
- *Structure*. Structure is one of the five literature themes used in this study. Structure refers to the established rules, goals, and norms of a business (Aurini, 2012; Bolman & Deal, 2017; Coulter, 2013; Gostick & Elton, 2012).
- *Successful learning center*. For the purposes of this research study, a successful learning center is defined as one that has been in operation for five or more years and is profitable (Yang, 2016).

*Training*. Training is one of the five literature themes used in this study. Training refers to the instruction, guidance, and support given to employees in order to improve learning outcomes (Birmingham, Pechman, Russell, & Mielke, 2005; Raley, Grossman, & Walker, 2005; Schumann, Peters, & Olsen, 2013).

#### **Theoretical Framework (Summary)**

The theoretical framework for this study is a two-part compilation of a business plan approach compiled by a UNESCO team called *The Smartest Investment: A Framework for Business Engagement in Education* and the resource based theory of entrepreneurship.

The Framework for Business Engagement in Education is a guide for education businesses to create learning opportunities for students while identifying the business case. Part of the business model includes a three-step process for engagement in education. Step one, "Make the business case," refers to aligning education with long-term growth and business strategy. This involves fostering innovation, addressing operational risks, improving reputation, boosting employee morale and retention, and professional development. Step two, "Identifying activities," refers to realizing which activities benefit both business and education challenges on a local level. Lastly, step three, "Be smart," implies the use of best practices in education and business, including social engagement (UN Special Envoy for Global Education, UNICEF, United Nations Educational, Scientific and Cultural Organization, & United Nations Global Compact, 2013, p. 12). This framework is well suited to this study because it involves the integration of education and business strategies in order to assist a larger number of students and pave the way for new private education businesses.

The Resource Based Theory of Entrepreneurship is a theory that stems from resource based theory (RBT). RBT is a theory with four conditions that led to business success: diversity of resources, use of resources that limit competition, limiting imitation of resources, and using innovation and ambiguity to limit competition. RBT in relation to entrepreneurship boasts two more concepts in addition: recognition of opportunities and organization and analysis of resources. (Alvarez & Busenitz, 2001; Kellermanns et al., 2016). Similar to the Framework for Business Engagement in Education, the Resource Based Theory of Entrepreneurship also makes connections to the literature themes and sub-research questions found in the following section. This theory relates to the study through the use of innovation in order to sustain businesses and lead them to success. RBT provides insight for the use of innovative resources and practices for entrepreneurs of any type of business to create business success. This includes entrepreneurs of education businesses, such as private learning centers.

#### **Research Questions**

The overarching research question for this purposefully selected case study was the following:

 What strategies does one successful K-12 learning center for students with learning disabilities in Southern California use to sustain their business?

The sub-questions for this study were:

- A. How does one successful K-12 learning center for students with learning disabilities in Southern California use training to sustain their business?
- B. How does one successful K-12 learning center for students with learning disabilities in Southern California use relationships to sustain their business?
- C. How does one successful K-12 learning center for students with learning disabilities in Southern California use innovation to sustain their business?

- D. How does one successful K-12 learning center for students with learning disabilities in Southern California use structure to sustain their business?
- E. How does one successful K-12 learning center for students with learning disabilities in Southern California use customer development to sustain their business?

The sub-questions were created and chosen based on the information and five themes discovered in the literature review: training, relationships, innovation, structure, and customer development. These five themes, described in detail in the following chapter, guided the direction of the research and the data collection in order to explore strategies for learning centers to sustain their businesses.

### Limitations

The ability to infer data across the population was limited because the sample for this study was taken from one learning center in Southern California as a case study (Lunenburg & Irby, 2008). To generalize study results, participants from other learning centers for students with learning disabilities outside of Southern California would need to be included. Access to participants was also limited due to their work schedules. Tutors and staff members of the learning center did not work at the center every day and many of the employees are part-time. Willing participants may not have been available during the days and times agreed upon by the researcher and the owner or manager of the learning center.

In gathering interview responses, there may have been limitations in asking participants to recall past experiences, as well as concerns for truthful responding. Participants may have recalled events based on emotions rather than lived experiences (Fowler, 2008). Employees who participated may have chosen to only share information that reflects positively upon them and their place of employment. Participants may not have had the ability to answer all of the questions asked of them or may have chosen to refrain from answering questions. Participants may have also chosen to remove themselves from the study at any time.

Limitations may have also involved the ability for the study to be replicated due to the number and location of participants. This includes limitations of the replication of a case study based off of an example of one learning center. Lastly, time was a limitation based on participant and researcher availability.

### Delimitations

The main delimitation was the purposefully small sample of gathering data from one learning center in Southern California. With a singular case study, collecting in-depth data from one location over a course of time, the sample size remains small and may have not reflected the population of learning centers in Southern California as a whole. The learning center selected had a focus on students with learning disabilities, which means that there was purposeful exclusions of other centers that also catered to this demographic. The selection was made from centers that are currently performing well and are popular in their area. Another delimitation is that the literature themes discovered through research were what the researcher saw as the most profound themes to center the study around. Time was also a delimitation in that the researcher was looking to complete their data collection in a timely manner, over several months.

### Assumptions

The researcher had assumptions based on her own background as an educational therapist with a learning center for elementary and middle school students in Southern California. The researcher assumed that learning centers are run as small businesses and that they are continually working to improve their business. The researcher also assumed that learning centers are looking specifically to assist students academically and are beneficial for students outside of regular school hours. As an educational therapist and owner of her own learning center, the researcher is both an educational colleague and a competitor of other learning centers in Southern California. The researcher used reflexivity to her advantage in order to control biases. She did so by practicing non-judgement through her own biases and knowledge and used that knowledge to help inform the types of questions and observations that she felt were beneficial to exploring the strategies of the learning center involved (Creswell, 1998).

The researcher assumed that learning centers are used to help students improve their academics and that the learning center involved in this study will continue to stay in business throughout the study. The researcher also assumed that learning centers play a valuable role in education and that the theoretical frameworks researched for this study are applicable to sustaining and learning center business. An assumption was made that the instruments used for data collection were accurate. Lastly, an assumption was also made that the participants had a sincere interest and were motivated to answer any questions truthfully.

#### **Organization of the Study**

This research study is presented in five chapters. Chapter One includes an introduction to the topic. Chapter Two presents a review of the literature, which includes five themes: social network, education, management experience, financial capital, and environment. Chapter Three describes the research design, the type of study, and the approach. Chapter Four presents the study's findings. Chapter Five provides a summary of the entire study and a discussion of findings.

### **Chapter Two: Literature Review**

This chapter will present a review of the literature that is related to strategies used to sustain small businesses. As there is very little research directly related to the problem of the study, strategies to sustain learning centers for students with learning disabilities, this literature review takes a broader look at the literature themes discovered through reviewing the research on education businesses and the success of other types of small businesses. Research of education businesses includes schools, after school programs, learning centers, and research regarding educators. The related literature looks to assist in the identification of commonalities between strategies for education and other types of small businesses.

This literature review explores the two theoretical frameworks involved in the research: The Smartest Investment: A Framework for Business Engagement in Education and the Resource Based Theory of Entrepreneurship. The literature review continues by exploring the historical aspects of learning centers and then includes five key literature themes. These literature themes are all related to strategies that are designed to sustain businesses and include training, relationships, innovation, structure, and customer development. Each of these literature themes are described in detail by examining their empirical research. Sources for this literature review were found using Scopus and a variety of EBSCOhost electronic databases such as Academic Search Complete, Business Source Premier, and ERIC. Other sources for this review include several recently updated academic texts with themes that include the topics of education, leadership, and management.

### **Theoretical Framework**

The theoretical framework for this study involves two different theories detailing a mixture of business and education, and a theory of entrepreneurship. These two theories, A Framework for Business Engagement in Education and the Resource Based Theory of Entrepreneurship, relate to the intersection between business strategies and education. As a combination, they provide insight into methods and principles that may be used to sustain a business that specializes in education.

A Framework for Business Engagement in Education. The Framework for Business Engagement in Education (UN Special Envoy for Global Education et al., 2013) integrates theories of education and business. This framework was created as a collective initiative by groups from the UN, UNICEF, and UNESCO in 2013. The initiative was created to aid in solving the problem of a "global learning crisis" (p. 6) and states that over 57 million children of primary school age across the world are currently not attending school. Of note, part of the solution to this crisis lies in the private sector as opposed to entrusting all of education to resources and institutions with public funding. More specifically, private education businesses are seen as a key component of helping improve education overall. For this reason, the Framework for Business Engagement in Education was created to assist private businesses that are involved in education to better grow their companies by detailing and sharing business strategies specially targeted towards the needs and challenges of education.

The framework includes three main sections: the business case, aligning business and education priorities, and the three-part process for engagement. The "business case" (UN Special Envoy for Global Education et al., 2013, p. 12) is a small section of the initiative which introduces why education businesses are important. Specifically, this section details the

importance and potential impacts of growing education businesses. The section entitled "aligning business and education priorities" (p. 10) discusses how companies, whether directly or indirectly involved in education, may still have a sphere of influence that impacts education in meaningful ways. This sphere is constructed through their core business, philanthropy, engagement, and partnerships that each contribute to varying degrees. The last main section, "three-part process for engagement" (p. 12), is the main focus of the initiative. It is also the portion of the framework that is most relevant to and therefore incorporated into this study.

The three-part process for engagement was created for two main reasons: to benefit businesses and to realize educational goals. In order to have a positive effect on the largest number of businesses, the process was designed for nearly any type of business looking to be involved in almost any aspect of education. This makes it especially appealing and relevant for private education businesses such as learning centers. The three components of this process are as follows: make the business case, identifying activities, and be smart. The synergy of these three components is designed to lead to a successful integration of education and business.

Step one, "make the business case," is directly related to the first section of the Framework for Business Engagement in Education, entitled the "business case" (p. 12). Step one asks the participant to establish a reason to be involved in education that aligns with the goal of growing the business. To assist with establishing this reason, five components are offered as a means to benefit the growth of businesses pertaining to education: Fostering innovation in education, Addressing operational risks, Improving brand leadership and enhancing corporate reputation, Boosting employee morale and retention, and Developing capacity of future employees. Step two, identifying activities, refers to actions that can be taken that can be specifically matched with each of the five components detailed in step one. The first component, "fostering innovation in education" (p. 14), has broad scope and includes technology, tools, distribution, and methods. For example, new educational technologies make it possible to innovate the delivery and management of education at the level of an individual student in a classroom, an entire school district, or as a global phenomenon with hundreds of millions of users. Additionally, the rise of virtual reality field trips, smartphones, youtube, and online portals of teachers sharing and selling resources are just a few ways that teaching methods can be innovated, distributed, and applied at scale.

The second section, "addressing operational risks" (p. 18), includes educating management and staff on internal policies and the creation of policies that are morally sound, understanding and utilizing a safe and clean learning environment, and the respect and support of human rights. The third section, "improving brand leadership and enhancing corporate reputation" (p. 20), involves using marketing campaigns that highlight the importance of education, sharing educational activities and outcomes with the community, and encouraging continued education for future leaders. The fourth section, "boosting employee morale and retention" (p. 24), involves offering ongoing training and education for employees and encouraging employees to volunteer for educational causes. Lastly, the fifth section, "actions taken toward developing capacity of future employees" (p. 28), includes the creation of training programs for future employees, expanding opportunities by providing vocational training and support for youth, and collaborating with other educators to design and inform one another of new educational opportunities. Each of these actions work toward the goal of making the business case.

Step three, "be smart" (p. 12), involves understanding human rights and responsibly applying best business practices according to the needs of the business. The business practice recommendations are to be applied to the five components from step one. There are six recommendations included to follow step three: respect and support education as a human right, promote equity and equality, partner with experts, be outcomes-oriented and results-driven, align with government needs in education, and be sustainable and scalable.

The Framework for Business Engagement in Education has varying connections to the literature themes identified as education and training, relationships, innovation, structure, and customer development. Table 1 distinguishes which sections of the framework overlap with which of the five literature themes.

Table 1

Framework Sections	Framework Subsections	Overlapping Literature Themes
The Business Case	Builds sustainable societies Saves lives Promotes economic growth Strengthens the local workforce Expands business opportunities Boosts wages	Structure Structure Customer Development Training Training, Relationships Structure
Aligning Business and Education Priorities	Core business Social investment Philanthropy Advocacy Public policy engagement Partnerships, collective action	Structure Relationships Relationships Relationships Relationships Relationships (continued)

The Framework for Business Engagement in Education With the Overlapping Literature Themes.

Framework SectionsFramework SubsectionsOverlapping Literature ThemesThree-Part Process1. Make the business caseInnovation, Structurefor Engagement2. Identify activitiesTraining, Relationships3. Be smartRelationships, Structure

Note. (UN Special Envoy for Global Education et al., 2013)

In the section of the business case, four of the five literature themes overlap with each of the subsections. Building sustainable societies coincides with the literature theme of structure in that it refers to a set of guidelines to follow to relate a business to education in order to be sustainable. These guidelines include "critical thinking, communication, and problem solving" (UN Special Envoy for Global Education et al., 2013, p. 8). The subsection saves lives also refers to structure. The framework suggests that keeping employees, companies, and the general workforce healthy in a medical sense creates a stable environment. Promoting economic growth overlaps with the literature theme of customer development because it relates to reaching out to the customer base of students and increasing their educational opportunities. Strengthening the local workforce refers to how education, in general, of employees helps businesses to be productive and to grow, which overlaps with the literature theme of training. Expanding business opportunities fits with the literature themes of relationships and customer development. When businesses invest in education, it can lead to new business relationships and shareholders, along with a potentially new customer base. Lastly, boosting wages refers to the literature theme of the structure of a business. Higher levels of education leads to economic growth through wage increases, which positively affects businesses.

In the section of aligning business and education, two of the five literature themes overlap with the subsections. Core business refers to how business operations, including factors such as policies and human resources, can address challenges in education, which relates to the literature theme of structure. The next three sections, social investment and philanthropy, advocacy and public policy engagement, and partnerships and collective action, all relate to the literature theme of relationships. These sections refer to relationships and engagement with the community through volunteering, engaging in policy, and creating partnerships, like minded individuals and organizations.

In the section of the three-part process for engagement, four of the five literature themes overlap with the subsections. The section is split into three parts, with the first two parts in reference to a set of five guidelines. Fostering innovation in education means involving new technology and products to enhance educations, which directly relates to the literature theme of innovation. Addressing operational risks relates to the literature theme of structure in that it refers to carefully constructing internal company policies for best business practices. Improving brand leadership and enhancing corporate retention overlaps with the literature themes of relationships and structure. Relationships are formed by sharing activities and values with other educational organizations. Structure is created through strong brand awareness. To have brand awareness, a company and its leadership need to fully understand the rules, focus, and goals set. Boosting employee morale and retention refers to providing training for employees so that they learn and improve upon skills, which directly relates to the literature theme of training. Boosting morale also means encouraging employees to involve themselves in the community through volunteering, which boosts relationships. The last of the five guidelines for parts one and two, developing capacity of future employees, overlaps with the literature theme of training.

Developing employees' capacity refers to providing workplace training for individuals to improve the work environment and experiences for both new and future employees. The third part of the process, be smart, is mainly related to the literature themes of relationships and structure. According to the framework, being smart means aligning with government education standards, partnering with experts, and building positive relationships to improve the outcomes of companies and the education business in general. Being smart also means building accountability in a business in order to be results driven, which overlaps with the literature theme of structure (UN Special Envoy for Global Education et al., 2013).

**Resource Based Theory of Entrepreneurship.** The Resource Based Theory of Entrepreneurship, briefly introduced in chapter one, is a theory that was born from two components: Resource Based Theory (RBT) and entrepreneurship. RBT is a standalone theory, with roots on the focus of resources enabling organizations. The theory was first defined by Edith Penrose in 1959. Penrose defined RBT as the physical items bought or created by a firm, as well as the people involved in the creation or execution of them for their companies (Penrose, 1959). However, the theory, with all its current components, was formalized by Jay Barney in 1991 and is also known as Resource Based View. RBT claims that the strategic resources of a business are what give it a competitive advantage over other similar businesses. The theory suggests that competitive advantages can be found and innovated from within a company using the resources available (Barney, 1991).

RBT is characterized by four learning objectives: diversity of resources, use of resources that limit competition, limiting imitation of resources, and using innovation and ambiguity to limit competition. Diversity of resources refers to the importance of heterogeneity in tools used by businesses. By having varying types of resources, or resources that other businesses do not have or are not aware of, businesses have a better chance at succeeding in innovative opportunities. Use of resources that limit competition also refers to using resources in innovative ways that differ from other businesses. Limiting imitation of resources means to create innovative practices in such a way that business competition cannot replicate the same strategies. Lastly, using innovation and ambiguity to limit competition is directly related to limiting imitation of resources. If a business innovates new strategies to help them succeed, they can limit imitation of their strategies by being ambiguous about them, or not sharing specific details with their competition.

RBT also makes a distinction between tangible and intangible resources. Tangible resources are those are clearly seen with the ability to be quantified. Intangible resources are those that are not quantifiable, including aspects such as knowledge, skills, reputation, and culture. Both types of resources are necessary for businesses to follow RBT and hold a competitive advantage over others (Kellermanns et al., 2016).

Sharon Alvarez and Lowell Busenitz published an article in 2001 combining RBT with entrepreneurship to create the resource based theory of entrepreneurship. When RBT is transformed into resource based theory of entrepreneurship, all of the aspects of RBT still apply and two additional concepts based on entrepreneurship are added: recognition of opportunities and organization and analysis of resources. Recognition of opportunities refers to the fact that searching for and discovering new opportunities as an entrepreneur acts as a beneficial resource. The analysis of resources includes the entrepreneur's organization and combination of resources as a resource in itself (Alvarez & Busenitz, 2001). The Resource Based Theory of Entrepreneurship, similar to the Framework for Business Engagement in Education, has varying connections to the literature themes and sub-research questions identified as training, relationships, innovation, structure, and customer development. Table 2 distinguishes which sections of the theory overlap with which of the five literature themes, relating it to the literature researched for this study.

Table 2

The Resource Based Theory of Entrepreneurship With the Overlapping Literature Themes.

Theory Concepts	Overlapping Literature Themes
Diversity of resources	Structure
Use of resources that limit competition	Innovation
Limiting imitation of resources	Innovation
Using innovation and ambiguity to limit competition	Innovation, Relationships
Recognition of opportunities	Innovation, Relationships
Organization and analysis of resources	Structure

*Note*. (Alvarez & Busenitz, 2001)

The six concepts in The Resource Based Theory of Entrepreneurship overlap with three of the literature themes. The diversity of resources concept is related to the literature theme of structure. To have varied resources, the owner of a business must develop a plan to acquire and use the resources. In order to develop such a plan, goals must be set, which refers to the structure of a business.

The next concept, the use of resources that limit competition, is directly related to the literature theme of innovation. To use resources in new and different ways than others, business

owners and staff need to be innovative in their thinking and actions. Doing so can lead to a competitive advantage over others.

Limiting imitation of resources also relates to the literature theme of innovation. Similar to the previous concept, this concept requires a business needs to think and act in inventive ways. A business is innovative and strategic by limiting the ability for someone else to copy what a resource is and how the resource is used.

The concept of using innovation and ambiguity to limit competition relates to the literature themes of innovation and relationships. The use of innovative practices directly relates to the literature theme of innovation. This concept also correlates with the literature theme of relationships. Associations with other individuals and businesses is beneficial to business success; however, by remaining ambiguous about new and innovative practices, a business can be the first to implement a practice and gain a competitive advantage. This means that business owners should be prudent about the intensity of their relationships with other companies as to not overshare information.

The following concept, related to the entrepreneurial side of RBT, recognition of opportunities, also relates to the literature themes of innovation and relationships. An entrepreneur must be innovative to recognize and invent new business opportunities. Also, through relationship-building with other similar companies, an entrepreneur can analyze their business practices and use them to help inform new business opportunities.

The last concept, also connected to the entrepreneurial side of RBT, organization and analysis of resources, is related to the literature theme of structure. Staying organized and efficiently analyzing resources and business opportunities means that an entrepreneur and business owner must set a plan and goals for their business. Setting a concrete plan creates a sound structure for a business to operate smoothly.

Along with each of these concepts, the Resource Based Theory of Entrepreneurship relates to sustaining businesses, which can be beneficial in exploring strategies to sustain learning centers. By keeping the business practices of a small business innovative using the strategies provided by this theory, the business has the opportunity to sustain itself by continually changing and growing with the world around it in order to stay relevant. Along with innovation, building new relationships with similar businesses and recognizing new partnership and alliance opportunities may allow businesses to stay relevant and sustain their companies as well.

**Summary.** The Framework for Business Engagement in Education and the Resource Based Theory of Entrepreneurship provide guidance as business theories to help sustain various types of business. More specifically, the Framework for Business Engagement in Education focuses on the importance of education-related businesses. The framework is meant to influence the growth and spread of private education businesses as a means to the improvement of education. In contrast, the Resource Based Theory of Entrepreneurship is strictly a business theory that focuses on keeping a business innovative and competitive in order to reach new levels of business success. Through both theoretical frameworks differ, they are comparable in that they both strive to provide insightful business strategies that encourage businesses to sustain their practices.

# **Historical Context**

The concept of a learning center stems from the notion of supplementary academic education. This type of education includes all types of academic assistance outside of regular school hours in the classroom. Supplementary academic education includes a variety of public and private options such as test preparation from schools, after-school academic activities and programs, private tutoring, learning centers, test prep centers, and online tutoring options. (Park et al., 2016). When looking specifically at learning centers, there are a variety of choices for students in early education, primary and secondary school, college students, and post-collegiate assistance (Stahl & Henk, 1993). Learning centers, in general, are meant to focus on student academic needs with one-on-one instruction and individualized tutoring plans (Christ, 1971). Learning centers have the potential to benefit many types of students academically, especially those who may have fallen behind or students with learning disabilities (Park et al., 2016).

Students with learning disabilities tend to struggle in a general classroom in school systems. According to Cortiella and Horowitz (2014) of the National Center for Learning Disabilities, learning disabled students are the largest category of students who receive special education services. Students with learning disabilities often struggle to maintain a high level of academic excellence in the school system. Overall, these students earn lower grades and one third of them have been held back a grade level at least once. Students with learning disabilities have goals similar to the general student population upon graduating high school, including going to college and joining the workforce. However, these goals are difficult to manage and achieve, especially when the students are frequently placed in a general education classroom with little to no individualized instruction. These students would benefit academically from individualized help and may receive outside of school hours at learning centers.

The successful education of students at a learning center is beneficial for both student and business success (Holliday, 2012). Student education in general is an important topic - students who do not graduate high school are at a great disadvantage when applying for jobs and working towards being productive members of society (Amos, 2008). One way to help students reach

these educational goals and learn new academic strategies is through tutoring (Anoka-Ramsey Community College, 2007). Supplementary education, such as tutoring, is beneficial to student academic outcomes and has becomes an integral piece in the education system (Park et al., 2016). Student education in relation to tutoring leads to enhanced learning outcomes (Truuvert, 2014).

Tutoring, assisting students with their academics outside of the classroom to gain a deeper understanding of subjects, as well as boosting academic confidence and grades, is an effective method of teaching in learning centers. As an example, Holliday (2012) conducted a study to assess the effectiveness of tutoring and education at learning centers. This was examined through the results of tutoring and whether or not learning had occurred for each student involved. Participants were students receiving tutoring who were enrolled in community college science courses, along with their five tutors. The exact number of participants was not stated, however, each of the five tutors worked approximately twenty hours per week with various students. Tutors were given a set of operational definitions that defined student learning and were told to use that list to classify the effectiveness of each tutoring session. Data was analyzed using the Chi-square goodness of fit test based on student results with each tutor. Results showed that students gained a significant amount of knowledge at a faster rate from the tutoring sessions and that tutoring proved to be a successful form of education, especially as a way to benefit students who have fallen behind their classmates.

Private learning centers and businesses, those which provide supplementary academic options for students at a cost, have become increasingly popular. Park et al. (2016) adapted a table using data from the Programme for International Student Assessment (PISA). In the years 2003 and 2012, PISA calculated what percentage of fifteen-year-old students took part in

supplementary academic education or tutoring outside of regular school hours. PISA made this calculation for many countries, including the United States. Results showed that there was a substantial increase in supplementary academic education for many of the countries, especially in the United States. Specifically, in 2003 in the United States, just over twelve percent of fifteen-year-old students participated in supplementary academic education. In the year 2012, that percentage increased to seventeen percent.

These centers' popularity is rooted in the desire to boost student academics, as well as a way to cope with the disappointment in school systems (Amos, 2008; Meyer & Rowan, 1978; Park et al., 2016; UN Special Envoy for Global Education et al., 2013). When the needs of students are not fully met within regular school hours, especially for students with disabilities, outside help in the form of centers is a viable option for families (Park et al., 2016). Using learning centers, educators can individually help students reach their educational goals and learn new academic strategies through tutoring (Anoka-Ramsey Community College, 2007).

### **Literature Themes**

A multitude of peer reviewed sources were gathered from books and online databases containing journal articles and research studies. From those sources, five themes emerged that support the problem of exploring strategies to sustain businesses: training, relationships, innovation, structure, and customer development. The themes are analyzed through research involving education businesses, general small businesses, entrepreneurship, and business management. Each theme is detailed thorough empirical research related to strategies to sustain businesses similar to learning centers.

**Training.** In the context of a business, education or otherwise, training refers to the instruction, guidance, and support given to employees in order to improve learning outcomes

(Birmingham et al., 2005; Raley et al., 2005; Schumann et al., 2013). Employees benefit from training related to their professional duties and responsibilities before beginning to work at a certain job, in addition to the knowledge and skills obtained from having completed their academic education in a related field (Simpson et al., 2004).

In regard to education businesses such as learning centers, teachers and tutors both benefit from continued job education (Hord, 1987; Radford et al., 2015; Sadioglu et al., 2013; Schumann et al., 2013). The growth of teachers' knowledge, including professional development and continually staying up to date in their understanding of educational strategies, leads to continued innovation and improvement. This betterment aids students and businesses alike in achieving positive outcomes including academic excellence and business growth (Hord, 1987). When teachers are educated about students' learning needs and taught specific skills to assist them, positive results are observed (Radford et al., 2015; Sadioglu et al., 2013; Schumann et al., 2013). More specifically in relation to learning centers, tutor education leads to student and business success as well (Bell & Mladenovic, 2008; Truuvert, 2014; Valkenburg, 2010). In order to educate management and staff, on the job training can be provided through various mediums, including peer and coworker observation (Bell & Mladenovic, 2008; Truuvert, 2014).

One well-established method of tutor training and education is through peer observations. Bell and Mladenovic (2008) studied peer observations for teachers at the University of Sydney where the university's program for tutor development conducted the study. The researchers explored tutor development through the school's academic program to understand the efficacy of peer observations. Of the 160 tutors in the program, 52 tutors elected to take part in the study and partner with another tutor for observations. Of the initial 52 participants, end results and data were collected from 32 of them. These partnered tutors were given instruction to observe one another and provide feedback during tutoring sessions for one semester. Peer observation forms and an end-of-semester survey were sources of qualitative and quantitative data. This form, which the participants types up and sent in after class, was based on a student feedback form that the university previously used to measure the success of the tutors and their teaching sessions. After completion, each tutor was given the feedback form completed by their peer tutor partner. The tutors were then given the opportunity to meet as a group and discuss their experiences from the study with an academic assistant available for discussion guidance and resources for each tutor.

The results from Bell and Mladenovic's (2008) study from the peer observation forms depicted a collection of mostly positive feedback for their peers. Results also showed that the majority of participants found the observations to be valuable and 88% of participants planned to change their teaching as a result of the observations and feedback. When the peer observers were asked about areas of improvement, their answers fell into into one of four categories: aiding students in understanding material, student participation in sessions, presentations skills, and the structure of sessions. Following this study, peer observations were adopted as an effective part of the tutor development program at the University of Sydney.

Continuing with peer observations, Truuvert (2014) wrote a journal article for Studies in Higher Education discussing a program he developed to advance the learning and capabilities of tutors. The tutors involved in the study were assisting with an undergraduate class in finance at Macquarie University in Australia. Similar to the research at the University of Sydney, the main focus of the program was to include outside observers and peer observation of tutoring sessions. It also involved third-party feedback, self-reflection, and feedback from students that was collected through surveys. These written surveys were given during class time in their final lecture of the semester by the faculty at the university. The program was meant to assist in the consistency of the information that tutors relayed to their students. This helped to ensure that students were given the necessary tools to comprehend every portion of the lecture material for their finance class. After the learning program was introduced, students reported improvements in their tutoring sessions and the tutors improved in their teaching skills and knowledge. Tutor improvement was made clear through the student feedback stating that their tutors were more helpful in the provision and understanding of the material, along with giving practical goals for students to achieve. Students reported that the course goals and standards were made clear through the tutoring process and that the material was fully covered. Overall, the study showed a notable improvement in the learning experiences for the students due to the training of their tutors.

Similar to training educators, training also benefits the management and staff of small businesses (Belitski & Heron, 2017; Esterhuizen, Schutte, & Du Toit, 2012; Fiet, 2001; Huang & Cho, 2010; Oana, 2012; Raley et al., 2005; Simpson et al., 2004; Solesvik, Westhead, & Matlay, 2014). Job training for students, educators, and employees can contribute to a healthy and stable organization. When employees are properly trained, they have the knowledge and skills to positively contribute to the operation and growth of a business. Staff training, in general, through professional development, job training, and continued education, can encourage business success (Birmingham et al., 2005; Raley et al., 2005). On the job staff training has an overall positive effect on the ability for businesses to sustain and grow (Simpson et al., 2004).

Simpson et al. (2004) used grounded theory to conduct a research study to better understand success through education, training, and experience for employees in small businesses. They used an inductive, systematic approach with qualitative data and gathered information from fourteen small business owners through semi-structured interviews. The owners of these businesses had anywhere from one to eighteen employees and were located in Sheffield, England. Results were gathered from the answers given during the interviews with the business owners. Each business owner had a different definition of success, meaning that each of their businesses were at varying levels of success in relation to sustaining businesses. The varying definitions of success included expansion and profit, workplace enjoyment, a feeling of achievement, and success through recognition and accolades. Results of the interviews showed the business owners felt that prior knowledge, having training and formal education in the past, was an important factor in small business success. Additionally, the training and education of employees positively affected each business' success as defined by their individual definitions. Though the results, which showed varying opinions on the definition of success for small businesses, each business owner felt as though varying forms of education and training for employees was effective for each of their businesses and related to the success of their businesses.

With respect to management and business owner training, small businesses are formed by entrepreneurs, many new to the world of business, who may benefit from education and training tailored specifically to entrepreneurs. Entrepreneurial education refers to educating and training aspiring business owners about the intricacies of entrepreneurship. Aspiring entrepreneurs benefit from to understanding what it means to be an entrepreneur and how to effectively open, run, and grow a business (Belitski & Heron, 2017; Fiet, 2001; Sánchez, 2013; Solesvik et al., 2014).

Sánchez (2013), a researcher and educator from Spain, conducted a study on an entrepreneurship education program to see if it was beneficial for students who may want to be entrepreneurs and start their own small businesses. The program was added as an elective class to secondary schools in Spain. Students who took part in the study were given a pre-assessment questionnaire of their entrepreneurial inclinations, which included rating their feelings toward self-efficacy, proactiveness, risk, and intention toward self-employment. The participants in the study included 710 secondary school students who were given the option of taking the elective entrepreneurial class. Prior to taking the classes and discovered through the results of the initial questionnaire, there was no significant difference in the entrepreneurial inclinations of the students in both groups. Even though all of the students had a similar level of interest in entrepreneurship, just over half of the students chose not to take the elective and were in the control group, and just under half of the students chose to take the elective entrepreneurial program and were in the experimental group. The entrepreneurship education program, offered over eight months, involved education about financing, marketing, accounting, and managing a business.

After the 8 months ended in Sánchez's study, results were calculated based on a questionnaire comparable to the one given initially. Results showed that the students involved in the experimental group, the ones who chose to take the entrepreneurial elective, became more interested in becoming entrepreneurs with raised scores in self-efficacy, proactiveness, risk-taking, and their inclination toward self-employment. The control group, the students who did not choose to take the elective, showed no difference in their questionnaire ratings. This shows that having exposure to entrepreneurship education and training early on had a measurably

positive effect on student interest and the knowledge needed to enter the workforce as entrepreneurs.

Solesvik et al. (2014) studied three universities in the Ukraine to understand the position of entrepreneurship in education. The study involved providing students with entrepreneurship education in order to understand the impact that the education had on the students' intentions for future career opportunities. The researchers gathered survey information from 321 students involved in either bachelor or masters business programs at the universities. Entrepreneurship education was provided through seminars and lectures. The surveys were structured questionnaires and were given to students in the control group (those who did not take the class) and students who received the entrepreneurship education. The results were very similar to Sánchez's study in that the students benefitted from entrepreneurship education. They showed that students who participated in entrepreneurship education and training had a higher intention of setting out as an entrepreneur and creating their own small businesses or forming new roles in small private or public companies.

Another form of training entrepreneurs is through mentoring. St-Jean (2012) researched mentoring used as a professional development method to train and assist new entrepreneurs so that they may be successful in their chosen careers. Research was completed and participants were chosen from a business related mentor network in Québec, Canada. The network, available for beginner entrepreneurs in business, provides mentoring across the province of Québec. The participants included everyone who was taking part in the mentor network program at the time of the study. There were 981 individuals in the program, and 360 of them chose to take part in the study. The participation sample was approximately half make and half female. Each participant was paired with an available mentor, with about 80% of the mentors being male and the rest

female. Of the participants, just over half obtained a college degree and ranged in age from young adults to middle and older adults. Over half of the participants had less than 5 years of experience in their chosen field, and 24% did not have any prior experience in their chosen field. The participants were given an online questionnaire in French that was later translated to English. The results of the study showed that the most effective factors of the mentoring process were how the mentor's career related to that of the mentee, as well as a psychological connection and their ability to serve as a role model for the mentees. In the mentoring process, the pairs who felt most successful built trust between one another. Trusting a mentor is a critical step to taking advice and learning from strategies. When two individuals in a mentoring scenario trust one another, they have confidence in each other and are more willing to be guided into new learning situations.

A study conducted by Cunningham, Huchting, Fogarty, and Graf (2017) proved another form of professional development to be successful. Their study was conducted at a Catholic PK-8 school where a professional development program was used to train teachers. The purpose of the study was to analyze the professional development program to understand its impact on educator classroom instruction and student engagement. The school, following a policy change, began admitting students with moderate disabilities. The teachers at the school needed assistance in understanding how to implement inclusivity, teaching to all different types of students, in their classrooms. Therefore, a professional development program was formed as an opportunity to train the teachers by giving them the information they needed to work with every type of student. The professional development model used was designed from a framework created by Universal Design for Learning Center for Applied Special Technology (UDL). The training program included rounds of one-on-one teaching sessions with the educators. The sessions were led by academic coaches from UDL. Prior to their training sessions, the teachers at the school had little to no experience teaching students with disabilities. The principal of the school and eight of the classrooms teachers participated in the study. All of the educators involved had an average of ten year of teaching experience.

The data was transcribed and coded qualitatively, and the coded data was memberchecked by the coaches at UDL. The data analyzed from this study was taken from the one-onone sessions and observations in the classrooms at the school. Semi-structured interviews took place during the one-on-one sessions as well. Interviews also took place prior to and after the professional development program was implemented. The results of the study found that both classroom instruction and student engagement improved after the professional development program was implemented. The teachers and staff at the school approved of the professional learning that took place and expressed their interest in further training and learning opportunities for their professional growth.

**Relationships.** Relationships involve aspects such as networking and making a connection with individuals through varying mediums both digitally and in person (Bolman & Deal, 2017; Epstein, 2002; Jochims, 2016). Building relationships with others through partnerships, networking, and collaboration allows for meaningful connections between businesses and their communities (Epstein, 2002). Through networking within a community, small business owners can advertise their business as a way to sustain and increase revenue. The community may include clients, families, potential employees, and other related businesses (Gerhardt et al., 2014). Networking and collaborating with other businesses and individuals can be used as a marketing technique to inform others about business goals and practices (UN Special Envoy for Global Education et al., 2013).

Gerhardt et al. (2014) conducted a study about ways to increase revenue for small businesses based on marketing through networking. They chose to research an example of a successful franchise, McDonald's. This popular franchise focused on networking with the community around them through marketing to boost sales, using a three level approach that included national, regional, and local advertising strategies. The strategy at the national level involved establishing partnerships with other large, national businesses. The regional level involved all regional franchise owners meeting and collaborating in groups to plan regional advertising campaigns. Lastly, the strategy at the local level involved networking with companies and individuals in their local communities as a way of advertising. Gerhardt et al. applied what they learned from the McDonald's franchise to networking through marketing strategies for small businesses. The researchers state the importance of building relationships at the regional and local levels, depending on the size of the small business. They concluded that the most important form of advertising involves individual owners networking within their community and advertising locally to increase revenue for their businesses.

Networking with individuals outside of a company can be an effective business strategy with additional beneficial outcomes other than those from networking with coworkers or team members. Keeping an open line of communication, understanding, and sharing ideas with other businesses can be a source of the innovation of better business practices (Epstein, 2002; Lencioni, 2012; Rese & Baier, 2011; Zapalska et al., 2015). Organized methods of communication and a clear and strategic plan of action are integral components of properly networking with other individuals. For example, networking and sharing ideas with individuals in similar fields of business has proven to be a popular strategy of business growth and organization (Rese & Baier, 2011; Zapalska et al., 2015). Additionally, networking to grow a

business can be successful when a business understands the reasons behind collaborating with others and how they plan to do so (Rese & Baier, 2011).

A large part of business relationships involve networking, trading knowledge, and gaining resources through connections with others, both inside and outside of a business. Sharing ideas with others is an important step of networking that can help achieve business success (Bond, 2012; Epstein, 2002; Esterhuizen et al., 2012; Hord, 1987; Jochims, 2016). Sharing information within a company is necessary for the company to function at full capacity, otherwise communication can become lost among individuals. This is especially true in terms of sharing information about the structure, the mission and rules, that a company follows (Jochims, 2016).

Along with networking, another factor of relationship building is involving the community surrounding a business (Epstein, 2002; Hord, 1987; Paluta et al., 2016; Yang, 2016). By keeping the surrounding community involved, especially families and other businesses with common goals, a business can share ideas and gain perspective from those around them. All persons who are stakeholders in the business, such as families, clients, surrounding businesses, and neighbors should stay involved (Epstein, 2002; Paluta, Lower, Anderson-Butcher, Gibson, & Iachini 2016).

In the primary and secondary education businesses, building relationships with the community includes building a bond with the parents of the students involved. Looking further into relationships, Paluta et al. (2016) examined the relationship between the quality and the outcomes of after school programs that were a part of the 21st Century Community Learning Centers (21st CCLC) program. The 21st CCLC model, authorized by the United State Congress in 1994, was created as a way to provide more support and opportunities for students and their

families. This model involved several types of programs including youth development, academic enhancement, and support programs for parents. Data was collected during the 2012-2013 school year from 405 21st CCLC programs in the midwest of the United States. Each of these programs served students in elementary, middle, or high school. Just over half of the programs served elementary school students. These programs were operated by community partners, such as the YMCA, by the schools or districts, and through education centers, cities, and universities. Halfway into the school year, the 405 21st CCLC programs were sent a survey as an evaluation of their programs. This online questionnaire took approximately twenty minutes to complete and paper copies were also provided as needed. Results showed a weak relationship between perceived program quality and outcomes. The strongest correlation was the relationship between family involvement and outcomes, which displayed the importance of building family relationships within the education community. The outcomes included the perceptions of student development, academic learning, and parent and family engagement. These results point to the need for positive relationships between educational programs for young students and the families of those students.

**Innovation.** Innovation is the creation of new ideas, concepts, structures, and resources in order to have a unique business that provides services unparalleled to others (Bond, 2012; Omri et al., 2015; Petersen, 2014; Rese & Baier, 2011). Innovation means to create something either completely new or an improved version of something already in existence (Omri et al., 2015). The theme of innovation focuses on creative people, practices, and resources.

Esterhuizen et al. (2012) state that innovation is essential for businesses to stay competitive and grow financially. The researchers break down innovation into two categories: product and process. Innovation of products involves developing tangible services. Innovation of process involves intangibles including new ideas, improvements on current ideas, and business procedures. Along with these two categories, the researchers note the idea of strategy innovation. This idea places a focus on strategic implementation of creative business ideas in a long-term manner, giving companies a competitive edge and chances for innovative success.

Individuals can be innovative by introducing new ways of thinking to enhance a business. Petersen (2014) wrote an article for *Education Next*, a journal about educational policy, in which she studied three education entrepreneurs in order to understand what made them successful and how their success can inform future innovative endeavors in education. Petersen focused on three successful education entrepreneurs: Larry Berger, Jonathan Harber, and Ron Packard. Each of these entrepreneurs found success through innovative uses of technology in the business of education. Petersen looked at what made these entrepreneurs successful and how their success may inform innovation in future educational endeavors. Larry Berger formed a company called Wireless Generation. In the year 2000, he was an educational innovator by incorporating technology in schooling in an unfamiliar way. Technology in education at that time was mostly focused on instruction, whereas Berger's company, Wireless Generation, focused on the needs of school and teachers to organize data and customise the curriculum. Jonathan Harber, similar to Berger, founded SchoolNet in 1998. SchoolNet, a tool for educators and school districts, gathers data to support academic improvements for students and to improve teaching and learning outcomes. Harber and Berger both found innovative ways to provide schools with educational support through technology. The last educational entrepreneur discussed in Petersen's article is Ron Packard, who founded K12 in the year 2000. Packard, a former investment banker, created K12 as a virtual school for both online and blended learning for primary and secondary school

students. K12 was a company that was innovative for its time and fulfilled a need for more innovative learning options that can be customized for students.

From her research on each of these entrepreneurial individuals, Petersen (2014) summarized what we can learn from their success in order to understand future innovation in education. Each of the entrepreneurs had a scalable, positive impact on the education community through patience, creativity, and adjustments. A commonality between the entrepreneurs is that they each were relentless in their task to solve an intellectual problem and were inventive in their solutions. Another commonality Petersen found is that these successful education companies created, along with others in similar positions, discovered a new way to gain revenue through technology and innovation. This differs from the common way through the state and district school systems, where most K-12 companies are situated.

Other than innovation through technology, businesses can find innovation through other types of resources. Omri et al. (2015) define innovation as "newness" (p. 1082). For businesses, this includes factors such as introducing new products, new methods and procedures, and new marketing strategies. The researchers conducted an empirical study to determine factors that affect the success of small businesses. They used qualitative and quantitative methods to gather their data through questionnaires given to their participants, 228 Tunisian businesses. The researchers first surveyed the businesses in the years 2010 and 2011. Then, using the data gathered from the surveys, questionnaires were created discussing human, social, and financial aspects of the businesses. More specifically, the researchers asked about innovation. Questions about innovation included asking about new products, new versions of products, new supply sources, new ways to manage finances, and new structures, systems, and procedures. The researchers

40

discovered that small businesses with great accessibility to financial and human resources are more likely to be innovative in their businesses practices, which can be a source of business success. Financial capability can assist in innovation by being able to produce new products and new marketing strategies. Having human resources and a collaborative team means that innovation can happen through new business methods and procedures.

In further exploration of innovation through management, Rese and Baier (2011) conducted research involving success factors in small and medium size businesses through innovation management in Germany. In this study, innovation management is described as tangible and intangible tools that enable management and employees of a business to work together under a common goal. The researchers compared success factors proven empirically with traditional factors discussed by the businesses. The empirical factors were network-related and involved networking through team organization and cooperation. The traditional factors included aspects of marketing and technology. In gathering participants, 271 German firms and research institutes received and returned a questionnaire in both the summer and fall of 2005 discussing network-related success factors. The participating firms were ones that developed innovative products, creative in either process or procedure. The managers of each of the firms and research institutes were the ones to respond to the survey. Results showed that along with traditional factors of success working for these businesses, networking was also an equally important factor to maintain an innovative environment through small business management. Each of these factors were shown to support the businesses by aiding in smooth operation and providing potential opportunities for expansion.

Innovation has a unique place in small businesses and entrepreneurial ventures in that there is more room for flexibility and creativity. This is due to the smaller business size and the opportunities to recreate the structure of a business regarding aspects such as goals and customer development (Bush, 2016). Oftentimes, innovative opportunities are not sought out by businesses due to the fact that they tend to focus on their current resources and guidelines instead of venturing out to create new ideas and visions (Hsiu-Li Chen, 2005). By embracing an innovative mindset, educators, business owners, and entrepreneurs can set out to create positive changes in new and competitive ways.

**Structure.** Structure includes the established rules, goals, and norms of a business (Aurini, 2012; Bolman & Deal, 2017; Coulter, 2013; Gostick & Elton, 2012). The theme of structure discusses the importance of clear, shared business norms that integrate well with one another. Bolman and Deal (2017) discuss the importance of the structure of a business in their book, Reframing Organizations. They reference the four frames of an organization as the structural frame, the human resource frame, the political frame, and the symbolic frame. The structural frame is viewed through structure and order, reward and punishment, and following rules based on company handbooks, contracts, and other important documents. The human resource frame is centered on human interaction, interpersonal relationships, and managing relationships. The political frame involves decision making, uses the power of politics, negotiation, and bargaining to achieve the desired results. The symbolic frame embodies true meaning through belief, faith, mission, and values. This frame involves the inclusion of events in a business such as rituals and ceremonies as a way to make members feel included. Though each frame helps to build the structure of a business, the structural frame is most directly related to the literature theme of structure. When changes are made through the structural frame, it means that adaptations are made through rules and regulations. Ensuring a solid structural frame of a business means placing an emphasis on productivity and standards, along with having clear goals and tight coordination through teamwork and understand roles in a business. Bolman and Deal believe that people are generally goal oriented. By having clear, easy to understand, measurable goals, the structure of a business is clearer and employees are more willing to work towards their goals.

Structure can be beneficial in both generic small business environments and educational business environments. When there is a lack of structure, the unclear business arrangement can lead to a stressful work environment as shown by a study conducted by Norton (2002) looking into learning center managers' job stress. Norton stated that the stress of center managers is comparable to the stress of other employees in academic roles and institutions, with several unique differences. While learning center managers do not necessarily have the stress of working with faculty members, the managers do have the stress of taking on many types of roles at one time. These roles may include teaching, accounting, office management, and advertising. Norton looked to identify the sources of stress for managers, contributing stress factors, and strategies to cope with the stress. The researcher gave 60 participants The Job Stress Survey, created by Vagg and Spielberger in 1999, during the spring semester of the year 2000. The participants varied in age, demographic, and levels of management experience. Ages ranged from young adults to middle and older adults. The majority of participants held at least their master's degree, while twenty percent held a doctoral degree. As for levels of experience, time spent in their current position ranged from one to twenty-two years and learning center management experience ranged from one to twenty-six years. Results from Norton's study showed that the most severe stressor was problems with management supervisors, lack of support from supervisors, and negative relationships between coworkers. Norton suggests that improving functions and the structure of the workplace including tasks, pace, and the physical environment, would enhance

relationships and improve the learning centers. When the structure of a business is well-defined, employees have the opportunity to understand their work relationships with one another, which can alleviate stress.

For a company to have a cooperative coworking space and working structure, a shared set of social rules and norms should be recognized between all individuals involved (Jochims, 2016; UN Special Envoy for Global Education et al., 2013; Yang, 2016). When everyone on a team is working toward a common goal, the team can find it beneficial to recognize and support the rules and morals of the company, as well as human rights in general. Promoting equality and teamwork through shared social rules strengthens teams and businesses (UN Special Envoy for Global Education et al., 2013). Best practices and social rules for coworkers can be followed through understanding the goals of the company they are associated with (Jochims, 2016).

Another crucial way to enforce a strong business structure is to have the leadership of a business understand the environment. When a company is formed, the leaders and management involved need to be educated on the intricacies of running a business to understand the focus of their business and the rules given to employees (Oana, 2012). This includes being educated on leadership, knowledge management, and business operation strategies. To run a successful business, leaders needs to be aware of effective operating practices (Covey & Merrill, 2006; Esterhuizen et al., 2012; Ferdowsian, 2016; Gostick & Elton, 2012; Holliday, 2012; Lencioni, 2012; Oana, 2012; Simpson et al., 2004). For example, in a learning center, administration need to be aware of best tutoring practices to effectively support students and to find success through positive student outcomes (Holliday, 2012). Even though prior knowledge and experience are beneficial to the success of a business, education and on the job training for management also have a positive effect on business success, including business growth and expansion (Simpson et al.)

al., 2004). Company leaders have the responsibility to acquire, share, and utilize management strategies through education and practice (Esterhuizen et al., 2012). Management strategies to ensure a solid business structure include but are not limited to education of organizational trust, focus and intention of business practices, and organizational health (Covey & Merrill, 2006; Gostick & Elton, 2012; Lencioni, 2012). Education about organizational trust focuses on understanding how to trust the opinions and suggestions of team members in order to successfully and smoothly work together (Covey & Merrill, 2006). In order to understand the focus of a business, leaders need to educate their employees about a clear mission and goals set to continually improve the business (Gostick & Elton, 2012). Through being educated about organizational health, business owners will have a better understanding of the morality and integrity of their businesses (Lencioni, 2012).

When all of these structural aspects are brought together cohesively, the result is called tight coupling. The concept of tight coupling is one stream of thought and action that fits well with the structure of a successful business. In reference to businesses, tight coupling means having an organizational structure with rules and norms that are mutually understood and are enforced through observation and management. This means that clear goals are established and rules are set that everyone in the business is aware of and can follow (Aurini, 2012).

Ferdowsian (2016) found success in tight coupling when he conducted a study addressing issues that prevent companies from sustaining success. Success, or excellence in this study, is defined by ten critical factors: products, financials, stakeholders, employees, leadership, societal, operational, innovation, alignment, and ethical excellence. The researcher gathered data from firms through a multiple case study. The firms involved included three groups that represented a variety of outcomes: firms who were recipients of various rewards related to success and

business excellence, Fortune-500 companies that have proven their success and responsibility, and firms with known ethical violations. Data was collected through the case studies, research, and field observations. Each of the firms were assessed based on an eleven point scale involving eleven different factors: meeting or exceeding the organizational excellence as defined by a variety of business award standards, following the criteria for being a socially responsible business, a compliant and ethical management approach, adding societal value and meeting the needs of an ever-changing business environment, delivering results with little to no internal conflict, providing a meaningful business purpose, providing professional development for employees, developing leadership that serve as positive role models, effectively collaborating with team members, demonstrating leadership and initiative, and displaying ethical and social responsibility. The firms were assessed based on each of these criteria through the three methods of data collection. Results from the data collection showed that to obtain business success, each of the eleven factors must play a role in the structure of a business. The results also highlighted the importance of the tight coupling, especially when relating to the categories of ethics and excellence.

**Customer development.** Customer development is focused on understanding the needs of a client base to better serve the population of a business (Bush, 2016; King-Sears, 2007; Yang, 2016; Zickel & Arnold, 2001). The theme of customer development is about understanding the needs of clients, putting the needs of clients first, and relating to a client base and gaining customers through marketing. Customer development in relation to education, specifically learning centers, includes tending to the needs of the client families and the students gaining academic support from the centers. In regard to learning centers, the staff must be educated about the needs of the clients to grow a customer base (Amos, 2008; Anoka-Ramsey Community

College, 2007; Holliday, 2012; Park et al., 2016; Truuvert, 2014). When creating a learning center business in which the clients are students with learning disabilities, the center should cater to the students' specific needs. In learning centers for students with disabilities, students must be educated both academically and socially about their own specific needs (National Joint Committee on Learning Disabilities [NJCLD], 2011; Merlone & Moran, 2008; Radford et al., 2015; Sadioglu et al., 2013; Valkenburg, 2010; Zickel & Arnold, 2001). Teaching students with disabilities how to self-advocate will help them become more successful and, in turn, create a successful business that students with learning disabilities can benefit from (Merlone & Moran, 2008; Zickel & Arnold, 2001).

Merlone and Moran (2008) conducted a study about the importance of students with learning disabilities understanding self-advocacy. The study involved fifth grade students with learning disabilities. The fifth graders were taught about learning styles and the intricacies of special education over a ten week period. Their lessons included information about strategies to cope with learning differences and how to advocate for themselves. Surveys were given at the beginning and end of the unit to assess student knowledge and attitudes about the subject. At the end of the ten week unit, students were assessed on knowledge gained from the course, asked what they found most important, and asked their opinions if the unit would be beneficial to other students with learning disabilities. The results showed that the students, in general, had a more positive attitude and better understanding about their disabilities after the unit. Initially, students felt as though their future career options were limited, but after the unit, their opinions made a positive change. The unit gave the students with learning disabilities more confidence in their abilities to learn and succeed, resulting in a positive academic experience. Along with giving student clients the tools they need to succeed in learning centers, students and businesses also benefit from educating tutors on strategies and teaching methods for working with students with learning disabilities (Merlone & Moran, 2008; Radford et al., 2015; Sadioglu et al., 2013; Valkenburg, 2010). One way for educators to help the clients of learning centers, in this case students with disabilities, recommended by the National Joint Committee on Learning Disabilities (NJCLD, 2011), is to give a comprehensive assessment as a way for teachers to understand student strengths and weaknesses to better individually assist their learning needs. Educators who have not been provided with strategies to help their students have been shown to have a negative reaction toward inclusion of students with disabilities. This negative reaction includes an unwillingness to learn new teaching strategies or adapt classroom lessons to fit the needs of students with disabilities. By understanding the needs of these students, teachers become more open and available for assistance (Sadioglu et al., 2013).

Customer service, in general, is about putting the needs of customers first. Bush (2016) completed dissertation research at Walden University detailing strategies affecting the ability for small businesses the sustain and be successful. The researcher gathered data through a qualitative, multiple case study to gain a better understanding of what makes small businesses long-lasting. Data was collected from six small retail businesses in Georgia through semi-structured interviews and field notes. Results of the data collection showed three key themes, one of which was supplying strong customer service. When the needs of customers are put first, a business can gain customer loyalty, leading to customer advertising, repeat business, and positive experiences. Customer needs may include factors such as personalized assistance, product or service individuality, competitive pricing, and ease of experience including proximity and process. If a small business does not choose to put customers first by catering to their needs, then

they will most likely have a much more difficult time selling their product and sustaining financially.

Yang (2015) also completed dissertation research at Walden University on the topic of key success factors for small businesses. The study focused on ways to identify strategies so that small businesses may successfully stay in business beyond five years. The research took place in Southern California and involved small retail businesses. The participants were two owners of small businesses in San Diego, California that had been in business for five or more years. The qualitative, multiple case study gathered data from the participants using semi structured, open-ended interviews. Results of the study founds themes including client loyalty and brand awareness, both of which are related to customer development. Client loyalty includes repeat business and strong customer relationships. Clients who trust a business and have seen positive results such as satisfaction of products and services, are more likely to advertise to their own network through word-of-mouth, increasing growth of a business. Brand awareness, increasing the visibility of a company to potential clients, helps build a client base. One can build brand awareness by targeting advertising to the needs of potential clients.

One way to target potential clients through advertising is to use social media to build online relationships with a customer base (Calderwood, 2009; Gerhardt et al., 2014). Social media is a successful online marketing technique that can inform a large, diverse group of people about a particular type of business in a timely manner (Gerhardt et al., 2014). Social media tools are not only beneficial for a variety of businesses, but can also be of great assistance to teachers and learning professionals (Calderwood, 2009).

## **Summary**

The review of the literature, including the theoretical frameworks, historical context, and the literature themes, serves as a foundation of exploring strategies to sustain businesses, namely learning centers for students with learning disabilities. The theoretical frameworks provided, the Framework for Business Engagement in Education and the Resource Based Theory of Entrepreneurship, are two business related theories that are meant to guide businesses through sustaining businesses in order to be successful entities. The Framework for Business Engagement in Education provides an educational perspective to business, while the Resource Based Theory of Entrepreneurship guides businesses in the direction of innovation and gaining a competitive edge. Through the guidance of these two frameworks and the literature reviewed, the five literature themes emerged: training, relationships, innovation, structure, and customer development.

In regard to the literature theme of training, employees have the opportunity to excel in their jobs through professional development and continued education, leading to business success. Employees who work for an education business, such as tutors, benefit from peer observations as a way of developing their job skills. Along with employees, leaders and entrepreneurs gain from training as well. When entrepreneurs are provided with specific entrepreneurial skills, they have a higher chance of starting their own businesses and taking on leadership roles. Also, when entrepreneurs are mentored by trustworthy, career-driven individuals, they are given the confidence to positive steps in the initiation of their businesses. On the job training for employees, business leaders, and entrepreneurs has a positive effect on business success. Relationship building is a vital component to the creation of a viable business and its expansion. The business should maintain positive relationships with all stakeholders which could include clients, families, neighbors, community members, and other similar businesses who are either in partnership or competition. For an education business, it is especially important for family members to stay involved in the academics of their students and the details of their educational paths. Attentive client relationships lead toward positive attitudes, client interest, and business success.

The third literature theme, innovation, allows for creativity and new products to emerge in businesses. Entrepreneurs in education have found innovative paths through creating new learning technologies, which set them apart from other K-12 ventures on their paths to success. Businesses with plentiful resources, financial and otherwise, have room for creativity and innovation to reach new levels of success. When these types of resources are made readily available, businesses have more options to innovative through testing out new ideas and new types of products. Another way to achieve innovation is through networking with other businesses and making collaborative connections. These connections may allow a business to grow their resources significantly and may also help provide the business with expansion opportunities.

The literature theme of structure shows how clear, measurable goals and well-defined roles leads to business success. Employees are ready to work toward goals that are measurable and can easily take action when guided in the right direction by management. Teams within a company better understand their work relationships with one another when the mission and goals are clear, which leads to a stress-free environment. Tight coupling, referring to having all of the structural aspects of a business aligned including streamlined norms and rules, is a beneficial way to successfully structure a business in order to sustain and grow.

Lastly, the literature theme of customer development plays a role in assisting business success through making customer needs a priority. A strategy of putting customers first for educational businesses such as learning centers is to keep customers well informed about their needs. For example, tutors should be attentive to the needs of students with learning disabilities by teaching them how to advocate for themselves in a scholastic environment.

The results showed that the students, in general, had a more positive attitude and better understanding about their disabilities after the unit. Initially, students felt as though their future career options were limited, but after the unit, their opinions made a positive change. The unit gave the students with learning disabilities more confidence in their abilities to learn and succeed, resulting in a positive academic experience. Customers, meaning students, families and a variety of business clients, support businesses financially and socially. By providing clients with attentive customer care, business owners may increase their brand awareness and bring new opportunities for business expansion.

The themes of training, relationships, innovation, structure, and customer development, along with the two theoretical frameworks, A Framework for Business Engagement in Education and the Resource Based Theory of Entrepreneurship, encapsulate the literature reviewed for exploring strategies to sustain learning centers and small businesses. These themes and frameworks will serve as a guide to inform my data collection from learning centers for elementary students with disabilities.

### **Chapter Three: Methodology**

Chapter Three provides a description of the methods for the study, including the purpose statement, research questions, research design and rationale, design validity, setting, population, sample, and sampling procedures, human subject consideration, instrumentation, data collection procedures, data management, data analysis, and positionality sections.

# Purpose

The purpose of this qualitative case study was to explore and describe strategies used by a successful K-12 learning center for students with learning disabilities in Southern California to sustain their businesses. This case study took place at one purposefully selected learning center catered to students with learning disabilities in Southern California that was currently successful in their ability to sustain their business. This qualitative case study incorporated data collection through interviews, observations, and archival data (Yin, 2014).

# **Research Questions**

The overarching research question for this purposefully selected case study was:

1. What strategies does one successful K-12 learning center for students with learning disabilities in Southern California use to sustain their business?

The sub-questions for this study were:

- A. How does one successful K-12 learning center for students with learning disabilities in Southern California use training to sustain their business?
- B. How does one successful K-12 learning center for students with learning disabilities in Southern California use relationships to sustain their business?
- C. How does one successful K-12 learning center for students with learning disabilities in Southern California use innovation to sustain their business?

- D. How does one successful K-12 learning center for students with learning disabilities in Southern California use structure to sustain their business?
- E. How does one successful K-12 learning center for students with learning disabilities in Southern California use customer development to sustain their business?

#### **Research Design and Rationale**

This study was a retroactive case study that, using a constructivist worldview, looked at archival data and ongoing practices of one learning center for students with disabilities in Southern California. A retroactive case study, looking at past and ongoing data, allows the flexibility to study each learning center on a timeframe that is beneficial to both the researcher and the center. This type of study also allows the ability to explore what has already made the learning center a successful business (Yin, 2014).

The constructivist worldview is defined as knowledge created socially (Creswell, 2014). Reality is up to interpretation and is discovered by individuals and groups. In this qualitative study, the focus is on gathering in-depth data measured by quality rather than quantity. Yin (2014) describes constructivism as an epistemology fitting for qualitative case studies, as qualitative researchers are also seen as interpreters. This means that researchers involved in qualitative case studies gather the knowledge around them to construct a viewpoint based on their investigation. The researcher is an educational therapist with her own learning center and is both an educational colleague and a competitor of other learning centers in Southern California. In line with the constructivist worldview, the researcher used reflexivity to her advantage. She practiced non-judgement through her own biases and knowledge and used that knowledge to help inform the types of questions and observations that she felt were beneficial to exploring the strategies of the learning center involved (Creswell, 1998). In this study, the researcher was seen as an interpreter who gathered knowledge about the learning center and used a variety of methods to construct a new viewpoint on strategies used to sustain the center.

### Setting

The study took place at one K-12 learning center for students with learning disabilities in Southern California. The learning center was located in an urban area in Southern California. The center served about 100 students total in grades K-12, but primarily catered to secondary school students. The students and staff were all English-speaking individuals.

The researcher traveled to the learning center four separate times over the course of two months to conduct interviews, observe their environments, and collect data. Four visits over a total of eight hours took place in order to allow enough time to gather the necessary data. The visits took place during normal business hours and after school hours while students were in tutoring sessions. This timing allowed for observation of interactions between management, employees, and client families. There were no observations of or interactions with children during the study.

# **Population, Sample, and Sampling Procedures**

The population for this study included all K-12 learning centers for students with learning disabilities in Southern California. The inclusion criteria involved learning centers that are located in Southern California, have a majority of English speaking individuals, work with K-12 students, and work specifically with students with learning disabilities. The sample was one purposefully selected learning center which caters to students with learning disabilities in Southern California that is currently successful in their ability to sustain their business. For the purposes of this research study, a successful learning center is defined as one that has been in operation for five or more years and profitable (Yang, 2015). The researcher reached out to K-12

learning centers for students with learning disabilities within Southern California. In order to recruit potential participants, individuals in leadership roles at the learning centers, including owners and managers, were contacted through emails and phone calls. A recruitment form was sent out to potential participants as well (Appendix A). A total of five learning centers were contacted, the last of which became a willing participant in the study.

### **Human Subject Considerations**

The researcher asked the operations manager and the owner of the learning center for permission to conduct the study at their learning center. The researcher reached out to the operations manager with a phone call. The discussion continued through the phone call, texts, emails, and an in-person meeting. At the in-person meeting, the operations manager of the learning center accepted to be a part of the study and the next steps were discussed, including permissions involved and beneficial meeting times. These permissions were be obtained in writing on company letterhead of the center (Appendix B). Identifying information of the learning center has been removed from the copy of permissions to protect privacy. Permission to have access to the learning centers and participants must be obtained from an authority figure at the center (Creswell, 2014). The learning center business name is not revealed to protect the privacy of the center and the employees. Employee names and personal information are not mentioned to protect their privacy as well. Informed consent forms were provided to the participating owner and the operations manager (Appendix C). Participants were asked to be interviewed, observed, and to share archival data of their learning center in order to explore strategies to sustain their businesses.

Participants in the study, including the business owner, operations manager, and other employees who took place in the interview process, were made aware of the option to opt out of the study at any time without penalty. Study participants were made aware of the minimal risks of the study through the informed consent form including possible fatigue, boredom, and discomfort sharing information related to their work environment. The owner and the operations manager of the learning center were made aware of the risks to the center. Risks include the fact that the learning center may be identifiable, even with a fictitious business, name based on the information given to the researcher for the study. This may put the center at risk if any negative information surfaces during the interview or observation process. A negative portrayal of any of the aspects of the learning center may affect business practices and hiring in the future. Also, proprietary information that is shared through the study may affect the learning center's competitive advantage.

The researcher did not record data in any way that may identify clients or families that are onsite at the learning center to protect their privacy. No information about specific students, their sessions, or their families were recorded. During the interviews, participants were given the option for breaks if needed and were notified of the right to refuse to answer any question. Both types of participants, managers and employees, were reminded that their answers are confidential and their participation in the study will not negatively impact their role at their learning center. Participants also had the option to have the study findings sent to them.

The records of this study, including all data collection and personal information, will be kept confidential. As per request, the audio files of the interviews were shared with operations manager for archival purposes. The operations manager secured and stored the audio files on his password protected computer at the learning center. The data is also stored on a password protected computer in the principal investigator's place of residence. A copy of Pepperdine's Institutional Review Board (IRB) approval is included (Appendix D).

# Instrumentation

Three methods of data collection were used to explore strategies used by one successful K-12 learning center for students with learning disabilities in Southern California to sustain their business: interviews, observations, and archival data. Three types of instrumentation were used in order to gain a substantial understanding of the learning center. Yin (2014) states that case studies benefit from multiple sources of data and that the variety allows the researcher to learn about and understand cases from multiple perspectives.

The researcher conducted an expert review of the data collection tools through an education colleague familiar with aspects of education learning centers, small businesses in general, and the small business management side of learning centers. This colleague reviewed the data collection methods to be used for this study, along with their details and the data collection plan. She agreed with the format of data collection and ensured that the essential details of sustaining a small business were incorporated in the plan for interviews, observations, and archival data.

The researcher followed preset interview protocols to conduct open ended interviews. Four participants were interviewed, including the operations manager, the chief of academics, the receptionist, and a new tutor. Interviews were selected to include both top-level management and employees serving in a tutoring role. The operations manager and the chief of academics, both tutors as well, received a set of interview questions directed toward their leadership roles (Appendix E). The other employees, including the receptionist and the new tutor, received a different set of questions regarding their roles at the center (Appendix F). These in-depth interviews will be open ended, leaving room for thick and rich descriptions. The researcher prepared the interview questions based on the literature themes previously discussed in order to

explore strategies used by each learning center to sustain their businesses, as shown in Table 3.

Table 3

Guiding Questions	Interview Questions	Literature Sources
RQ1: What strategies do successful K-12 learning centers for students with learning disabilities in Southern California use to sustain their learning centers?	<ul> <li>What is your role at [learning center]?</li> <li>How did you create the learning center? (If applicable)</li> </ul>	Bolman & Deal, 2017
Sub RQ A: Use of workforce training to sustain business	<ul> <li>What is your approach for hiring and training employees?</li> <li>Do you participate in professional development or training for work?</li> </ul>	Birmingham et al., 2005
Sub RQ B: Use of relationships to sustain business	• How do you manage relationships with customers?	Bolman & Deal, 2017; Epstein, 2002; Jochims, 2016
Sub RQ C: Use of innovation to sustain business	• What role does innovation play in the operation and growth of your learning center?	Bond, 2012; Omri et al., 2015; Rese & Baier, 2011
Sub RQ D: Use of workplace structure to sustain business	<ul> <li>What are the rules and protocols that you have in place for your employees to follow?</li> <li>What are the rules and protocols in place at the learning center?</li> </ul>	Aurini, 2012; Bolman & Deal, 2017; Coulter, 2013; Gostick & Elton, 2012
		(continued)

Guiding Questions	Interview Questions	Literature Sources
Sub RQ E: Use of customer development to sustain business	<ul> <li>How do you reach new customers?</li> <li>Do you measure client success and what role does that play?</li> <li>What steps do you take, if any, to ensure customer satisfaction?</li> </ul>	Bush, 2016; King- Sears, 2007; Yang, 2016; Zickel & Arnold, 2001

These in-depth interviews allowed for detailed descriptions and details with real accounts provided by leaders and employees of the centers. The open-ended interviews were beneficial to the study in that they provided thick and rich descriptions through the lens of the participants (Yin, 2014).

The researcher also collected observations through field notes to observe the daily operation and environment at the learning center, while keeping in mind the five literature themes found to help explore strategies to sustain learning centers: training, relationships, innovation, structure, and customer development. The researcher took detailed notes of these observations in order to qualitatively measure the business operations of the learning center. The researcher was given permission to take photos at the learning center. The photos were used to assist in recollection of environmental details. The method of observation helps the researcher gain a better understanding of the business environment (Yin, 2014). The results of the data gathered from observations may differ from that of the interviews due to the fact that observations are seen through the lens of an observer instead of an employee.

The final method of data collection, archival data, included records from the learning center data stored in physical files and on computers. The researcher kept track of the data found

using the archival data collection guide (Appendix G). The researcher prepared the archival data collection categories based on the literature themes previously discussed in order to explore strategies used by the learning center to sustain their business, as shown in Table 4.

Table 4

Guiding Questions	Archival Data Collection Categories	Literature Sources
RQ1: What strategies do successful K-12 learning centers for students with learning disabilities in Southern California use to sustain their learning centers?	<ul> <li>Learning center website</li> <li>Other available information on the internet</li> </ul>	Omri et al., 2015
Sub RQ A: Use of workforce training to sustain business	<ul><li>On-boarding</li><li>Professional development</li></ul>	Birmingham et al., 2005; Raley et al., 2005; Schumann et al., 2013
Sub RQ B: Use of relationships to sustain business	• Flyers/brochures	Bolman & Deal, 2017; Epstein, 2002; Jochims, 2016
Sub RQ C: Use of innovation to sustain business	• Types of technology used	Omri et al., 2015; Rese & Baier, 2011
Sub RQ D: Use of workplace structure to sustain business	<ul><li>Flyers/brochures</li><li>Learning center website</li></ul>	Aurini, 2012; Bolman & Deal, 2017; Coulter, 2013; Gostick & Elton, 2012
Sub RQ E: Use of customer development to sustain business	<ul><li>Flyers/brochures</li><li>Learning center website</li></ul>	Bush, 2016; King- Sears, 2007; Yang, 2016; Zickel & Arnold, 2001

Relationships among Guiding Questions, Archival Data Collection Categories, and Literature

The archival data collected included aspects such as website information, internet search results, information from the learning center's online blog, on-boarding, professional development, flyers and brochures, and information about technology used. Looking at archival data gives the researcher an understanding of the flow of business operations from the past to the present (Yin, 2014).

### **Data Collection Procedures**

Data collection began in early January 2018, soon after the owner and operations manager at the learning center gave written consent to take part in the study. The researcher scheduled appropriate days and times with the operations manager of the learning center to make in-person visits. Four visits to the learning center took place over the months of January and February 2018, each visit lasting a total of two hours to account for eight total hours of time spent at the center for data collection. The first visit involved a tour of the learning center and an introduction to the employees. During the second visit, center and employee observations and the collection of archival data in print form were collected. Lastly, the third and fourth visits involved the collections of the rest of the archival data and the four interviews. During each of these visits, the researcher spent an appropriate amount of time at the learning center to gather all necessary data.

In regard to interviews, the researcher was equipped with the lists of open-ended interview questions for management and employees, notebook paper, the archival data collection guide, and a recording device. The researcher interviewed four participants, including the operations manager, the chief of academics, the receptionist, and a new tutor. The interviews were selected to include both top level management and employees serving in a tutoring role. The interview questions were presented in the same manner for each of the participants in the study. Two sets of interview questions were used with one type of questioning for owners or managers and one type of questioning for employees. The operations manager and the chief of academics, both tutors as well, received a set of interview questions directed toward their leadership roles and the receptionist and the new tutor received a different set of questions regarding their roles at the center. The two sets of questions are similar and both follow a structure involving the five literature themes. The interview questions for the owners and managers ask about details regarding their leadership roles such as how their learning center began, how they acquire and train employees, and how they work with their client base. The employee interview questions focus on their contributions and how the employees view their work environment. The notebook paper was provided for the researcher for taking notes during the interview. The recording device was used to record each interview, which later was transcribed to a written record through the use of HyperTRANSCRIBE.

During the learning center visits, observations were made by the researcher in order to understand the environment and day-to-day operations of the learning center. The researcher made field notes with each of the five literature themes and sub-research questions in mind during the visits. The researcher also received consent from the operations manager to take photos. A camera was brought to the learning center to take pictures of the center in allowable locations to assist in later recollection of the visit. No clients or family members or employees were included in the pictures to protect their privacy. During the first visit, the operations manager gave the researcher a tour of the facility. During each visit, observations were also made by watching the communication and relationships between the management and employees, as well as employees with each other. Archival data was also collected during the visits to the learning center and over the computer through the use of internet searches and email exchanges between the researcher and the operations manager. During center visits, the researcher gathered print archival data including brochures and a sample of the employee timesheet. Through an internet search, the researcher gathered data through the learning center website, their blog, and results when typing the name of the center into an internet search engine. Through email, the operations manager provided the researcher with the teacher training manual and the operations manual given to the employees of the learning center. The researcher was given permission to collect this information through email and to store it securely on the private computer.

### **Data Management**

The researcher used HyperRESEARCH, Researchware, to analyze the data of the study, and well as member checking to ensure validity of the data collected. Outcome validity of the study was defined through triangulation of the data and interrater reliability. Triangulation is defined as combining various sources and methods of data collection in order to find common themes (Creswell, 1998). For dialogic validity, the researcher recorded the interviews using a personal recording device. The researcher will also take handwritten notes while recording as needed in order to follow the guide of interview questions. The researcher used HyperTRANSCRIBE, Researchware, to play back and transcribe the four interview that took place at the learning center. Once the transcription was complete, the researcher listened to the audio recordings while reading the transcriptions of the interviews to ensure accuracy.

The researcher obtained audio recording permission from all of the participants who were interviewed prior to collecting data. Those with access to the data include the researcher, interrater, and dissertation team. The operations manager was given access to the recorded interviews for the learning center's own archival purposes. The data will be destroyed five years after completion of the study.

#### **Data Analysis**

The data collected from the learning center through interviews, observations, and archival data was analyzed through HyperRESEARCH. The data was separated into three cases, one case for each data collection method including observations, interviews, and archival data. Each of the three cases included sources relevant to the data collection method. The observation sources included a document of all observations written by the researcher during the learning center visits. The interview sources included four separate documents for each interview: the operations manager interview, the chief of academics interview, the receptionist interview, and the new tutor interview. The last type of data collection, archival data, included the archival data sources found: website information, blog information, customer reviews, a tutoring session and class brochure, the employee timesheet, the teacher training manual, and the operations manual. Within each case, the data were coded into five code groups including the five literature themes: training, relationships, innovation, structure, and customer development. Through analysis of the data, inductive codes were found and categorized by each code group. Coding is an essential method of data analysis for qualitative research in that it breaks down detailed descriptions through structured themes (Creswell, 1998). Once the data was coded, the researcher compared the code lists of the three data collection methods in order to find potential commonalities. The researcher used the software to find common words and themes, thick and rich descriptions, and memorable quotes that relate to exploring and describing strategies to sustain learning centers in relation to the research questions and literature themes. Each group of archival data was compared with one another and common themes were accounted for.

Once these data were analyzed, the results were interpreted through a variety of means. The findings from the data were triangulated and grouped into the five literature themes: training, relationships, innovation, structure, and customer development. These deductive themes were born from peer reviewed sources and books that were presented in Chapter Two's literature review. The literature themes were used as code groups for the data analysis to understand how the findings of the study connect to the findings from the literature. Each of the five groups were compared and studied to find consistencies between the various types of data to establish validity.

After coding the data set for the deductive literature themes, the researcher began an inductive coding process and thematic analysis. Creation of themes was done by analyzing the data from the ground up by organizing the information into progressively more abstract components (Creswell, 2014). The researcher analyzed the coded quotations, notes, and varying types of artifacts to find commonalities in each of the five code groups. Common words, phrases, and concepts were discovered and shared amongst the five groups and further dissected to understand why they were prominent components of the learning center. These shared concepts were thoroughly analyzed and formed into five, peer reviewed themes. These themes are presented in chapters four and five. By triangulating the data gathered from interviews, observations, and archives, the researcher validates the findings by showing their overlap between the literature themes (Creswell, 1998). The findings are presented through the discovery of themes, thick and rich descriptions, and figures, tables, and charts of the data.

### Positionality

The researcher is an educational therapist with her own learning center for elementary and middle school students with learning disabilities in Southern California. The researcher has the opportunity of benefitting her own learning center through conducting research to explore and describe strategies to sustain learning centers similar to her own. The researcher has a vested interest in the success of learning centers in the area through connections to families, communities, and schools. As an educator, the researcher may have biases of her own related to how learning centers should sustain themselves. As a competitor to the learning center recruited for this study, the researcher will notify the owner or manager involved in the study of her current position and described the potential benefits for all parties involved in the study. To manage her biases, the researcher will follow a specific guideline of open ended interview questions that follow the literature themes. No leading questions will be asked of the participants.

### **Chapter Four: Report of the Findings**

Chapter Four provides a brief overview of the study, interviewee demographics, a description of the artifact collection, observations, and a data summary. This chapter also discusses the five sub-research question topics in relation to the data analysis: training, relationships, innovation, structure, and customer development. Lastly, themes discovered through the data collection are introduced. The themes are as follows: Theme 1: Engaging in closed-loop communication with all stakeholders; Theme 2: Taking a holistic approach to student improvement; Theme 3: Providing differentiated instruction for a personalized experience; Theme 4: Engaging in a growth mindset; Theme 5: Setting an intention for the learning center to follow. The chapter concludes with a summary of the key findings and themes.

# Purpose

The purpose of this qualitative case study was to explore and describe strategies used by a successful K-12 learning center for students with learning disabilities in Southern California to sustain their businesses. This case study took place at one purposefully selected learning center catered to students with learning disabilities in Southern California that was currently successful in their ability to sustain their business. This qualitative case study incorporated data collection through interviews, observations, and archival data (Yin, 2014).

### **Research Questions**

The overarching research question for this purposefully selected case study was the following:

1. What strategies does one successful K-12 learning center for students with learning disabilities in Southern California use to sustain their business?

The sub-questions for this study were:

68

- A. How does one successful K-12 learning center for students with learning disabilities in Southern California use training to sustain their business?
- B. How does one successful K-12 learning center for students with learning disabilities in Southern California use relationships to sustain their business?
- C. How does one successful K-12 learning center for students with learning disabilities in Southern California use innovation to sustain their business?
- D. How does one successful K-12 learning center for students with learning disabilities in Southern California use structure to sustain their business?
- E. How does one successful K-12 learning center for students with learning disabilities in Southern California use customer development to sustain their business?

## **Interviewee Demographics**

Four participants at the learning center were interviewed: the operations manager, the chief of academics, the receptionist, and a new tutor. Interviews were selected to include both top level management and employees serving in a tutoring role. Table 5 summarizes basic information about the participants who were interviewed.

Table 5

Participant Title	Time at the learning center	Gender
Operations Manager	2 years	Male
Chief of Academics	6 years	Male
Receptionist	3 days	Female
Tutor	1 month	Male

### Interviewee Information

# **Artifact Collection**

Along with interviews and observations, seven different types of artifacts were gathered from the learning center, including the website, blog, employee timesheet, class schedule flyer, operations manual, teacher training manual, and customer reviews. The website, blog, and customer reviews were easily accessible online. The employee timesheet and class schedule flyer were artifacts that were readily available at the front desk at the learning center. The operations manual and teacher training manual were two sources obtained by the operations manager and not available to the public. Each of these artifacts provided information about the learning center through the varying perspectives of customers, employees, and management. Table 6 displays each type of artifact, the locations they were acquired, and whether the artifacts were digital or print sources.

## Table 6

Artifact Type	Location Acquired	Digital/Print
Website	Search Engine	Digital
Blog	Search Engine	Digital
Employee Timesheet	At the center	Print
Class Schedule Flyer	At the center	Print
Operations Manual	Email	Digital/Print
Teacher Training Manual	Email	Digital/Print
Customer Reviews	Search Engine	Digital

# Artifact Information

## Observation

The learning center was observed on four different occasions, including two days in January and two days in February. Observations were collected through the use of photography to aid in recalling the physical structure of the center and hand-written notes taken in the researcher's observation notebook. Eight pages of notes were recorded in the observation notebook by the end of the data collection in February. Figure 1 is an example of one of the pages of hand-written notes during time spent observing the learning center, including information about the physical space and employee interactions.

Physical location of the learning center: - Two entrances, came in through back entrance each time + explored front entrance - Reception area accessed by front entrance t desk faces away from front door - "Waiting room" is a large classroom used as a study hall while students wait - A couple of mant offices @ back of center by back entrance - Long hallway leading to several smaller classrooms w/ #s on the doors - Small classrooms have blank walls, nothing on desks. Very bare! \* Ask why - Bookshelf @ end of Small classroom hallway wil books of all Subjects -can't stell how it's organized, but looks clean \* Take photos of hallway w/ books + classroom example Interactions - Receptionist smiles + greets everyone who walks by - Everyone acknowledges each other in passing

Figure 1. Hand-written observation notes.

# **Data Summary**

Once the data from interviews, observations, and artifacts were combined and coded using HyperRESEARCH, 22 new codes were developed and categorized into five code groups reflecting the sub-research questions: training, relationships, innovation, structure, and customer development. Each code group contains a set of codes found inductively by the researcher through the analysis of the data. At least three codes and up to six codes were inductively found for each code group, showing a strong connection between the data collected and the subresearch questions acting as code groups. Table 7 shows the code groups, each with the corresponding inductively developed codes found and their frequency within the dataset. Table 7

Code Group	Number of Code Tags in Group	Code Name	Code Count
Training	61	Orientation	18
8		Teaching Learning Styles	16
		Teaching Motivation	9
		Peer Observations	7
		Lectures	6
		Feedback	5
Structure	49	Location and Space	14
		Scheduling	12
		Hiring Process	9
		Vision	9
		Report System	5
Relationships	35	Making Connections	13
r i r i r i		Building Confidence	8
		Consultation	8
		Motivation	6
			(continued)

Code Groups and Inductively Developed Codes

(continued)

Code Group	Number of Code Tags in Group	Code Name	Code Count
Innovation	25	Technology Integration	12
		Embracing Change Neuroscience	7 6
Customer Development	25	Marketing Customer Reviews Discounts Student Improvements	9 7 5 4

The text from all of the interviews, observations, and archival data was also placed into HyperRESEARCH and the word counter tool was used to find the amount and order in which descriptive words were used. The descriptive words include all adjectives, nouns, and action verbs found in the data. Table 8 displays the top 25 descriptive words used as well as their word count in the combined texts.

Table 8

Data Word Count – Top 25 Descriptive Words Used

Word	Word Count	
Student/students	101	
College	37	
Tutor/tutoring	36	
Teaching/teachers	33	
Needs	28	
Class/classes	26	
Time	18	
Technology/tech	14	
Parents	14	
Self	14	
		(continued)

Word	Word Count	
Give	14	
Academic	14	
Study	13	
Learning	12	
New	12	
Change	11	
Education	11	
Confidence	10	
Care	10	
Motivation	9	
Different	9	
Difficult	9	
Science	9	
Available	8	
Provide	8	
Science Available	9 8	

The top 25 descriptive words used ranged from being in use in the data eight times to 101 times. In order to understand the importance placed on the descriptive words in the data, a word cloud shows a conglomeration of the words presented visually, showing their amount of use by the size of the word. Figure 2 shows each of these words in a visual representation of a word cloud.



Figure 2. Word cloud: Top 25 descriptive words used.

The descriptive words displayed in Table 8 and Figure 2 provide an example of the words and ideas that the learning center placed a great importance on through their manuals, center information, and their own words (Bernard, Wutich, & Ryan, 2017). The words student and students were brought up the most and to a large degree, showing that the learning center is student-focused and places student needs first, before the needs of teachers, tutors, and parents. Along with student and students, the highest counted verb is the word need, as the needs of the students come first.

#### **Research Sub-questions**

Each of the research sub-questions served as code groups, deductively used for data collection and entered into HyperRESEARCH prior to inductive coding. Those code groups are: training, relationships, innovation, structure, and customer development. This section reports the codes that were inductively found in the analysis of the learning center data that correspond with each of the research sub-questions.

**Research sub-question 1:** Training. Research sub-question 1 asked, "How does one successful K-12 learning center for students with learning disabilities in Southern California use training to sustain their business?" This question is paired with the literature theme of training and data collected in reference to this question was placed in the code group, "training." Within the code group of training, six new codes were discovered: feedback, lectures, orientation, peer observations, teaching learning styles, and teaching motivation.

*Feedback.* During their interviews, the operations manager and the chief of academics at the learning center both discussed the importance of getting feedback on tutor performance. The operations manager stated that the optimal way to receive feedback on tutor performance is by speaking with the students and their parents, and not by "self-report" of tutors assessing their

own performance. The learning center also has security cameras in each tutoring room for safety. If a problem arises, they have the opportunity to review the footage from the tutoring room.

*Lectures.* The chief of academics discussed training tutors through lectures at his interview. "Philosophical motivational speeches" are given, as well as teaching seminars. The chief of academics plans to grow the seminar program to give the tutors more opportunities to learn. In a separate interview, the new tutor discussed a meeting that he attended where he listened to information about teaching and where the company is headed in the future.

*Orientation.* The main form of tutor training at the learning center is called orientation. The center uses their teacher training manual to train tutors and the operations manual to train other members of the staff in leadership positions. The teacher training manual contains information about company rules, regulations, and expectations. The manual also provides information about teaching, classroom management, and a tutoring session breakdown, describing how to start, continue, and end an hour tutoring session. The operations manual trains the staff about rules and regulations, expectations, communication, and scheduling.

Along with the training manuals, when a new tutor begins working at the learning center, they are given a tour and they meet with an employee in a leadership role to discuss the training manual and any other pertinent information. When the receptionist was hired, she had a meeting with the operations manager about greeting others, handling calls, constructing conversations with students and parents, and leading with a positive attitude.

*Peer observations.* Peer observations also take place at the learning center to train new and current tutors. The chief of academics has tutors observe on his sessions with students and says that this type of observation is the "easiest method to teach them." The new tutor stated in

his interview that he sat in on a session with a tutor who has been with the center for a while to help him understand what is expected of him. During company meetings, tutors also share their personal experiences with one another so that they can learn from each other.

*Teaching learning styles.* The learning center trains their tutors by teaching them the importance of understanding each student as an individual through learning styles. The teacher training manual contains a section with resources about learning styles and teaching styles. The company website acknowledges the importance of learning styles and understanding "that every student is unique." The chief of academics discussed training about teaching styles in the form of understanding the specific needs of each student. When tutors begin working with a new student, the chief of academics aids them by sharing information about the student's specific needs and how to assist them. He stated:

I do psychological evaluations and counseling here, and because of [the students'] psychological issues, we have a lot of students that are not diagnosed for special needs but need special education care. I have their special stories ready for the individual tutors. They get individual care.

The new tutor also discussed how he was given information about his students before working with them. He "taught a student with autism" and was prepared with information about the student and his needs before teaching.

*Teaching motivation.* Teaching tutors about the power of motivation is a notion that appears numerous times in the teacher training manual. The training manual has a section about teaching motivation and how to be an effective teacher by motivating students. The manual urges the tutor to "find out what drives a student" because "without motivation, they will not retain the information."

**Research sub-question 2: Relationships.** Research sub-question 2 asked, "How does one successful K-12 learning center for students with learning disabilities in Southern California use relationships to sustain their business?" This question is paired with the literature theme of relationships and data collected in reference to this question was placed in the code group, "relationships." Within the code group of relationships, four codes were discovered: building confidence, consultation, making connections, and motivation.

*Building confidence.* One aspect of creating strong relationships found in various sources of archival data from the learning center is building confidence. The blog talks extensively about the importance of confidence building for students. Self-confidence can be achieved by praise and respect from authority figures. The center believes that confidence and academic success are linked and that helping students build confidence will guide them to achieving their academic goals. In one of the customer reviews about the learning center, a student felt successful because her tutor gave her confidence.

*Consultation.* Consultations are also a vital part of relationship building, according to the operations manual and the interviews with those in leadership positions. The operations manual has a full section on consultations, including managing relationships with parents, tutors, and their students. Before a student is enrolled, the chief of academics has a "minimum one hour [consultation] with their parent and the student to have an analysis" that acts as an academic and personal profile analysis. Parent consultations also take place during the student's enrollment and throughout their time at the center to discuss goals and any concerns that may arise.

*Making connections.* A key component of relationship building at the learning center is making connections between the tutors and the students. Several of the customer reviews stated that students made positive connections with their tutors and that it helped them learn the

material and succeed. Students were able to open up to their tutors and felt comfortable in the learning environment. The chief of academics stated that "the most important thing is the chemistry with the students." He shared that the goal is to make sure the students want to attend their sessions and if a student has not built a connection with their tutor, they will not want to continue. The operations manager looks at how the students engage with their tutors. On the topic of matching students with tutors, he stated:

It is completely up to us. We're a matchmaker, we are like Uber or Tinder for education. It's a coincidence of many things, of wants and needs. It is the teacher and the subjects they know. It's the students and the subjects they need to be taught. We bring them together.

*Motivation.* Building relationships through motivation is a part of the training manual, blog, and several customer reviews. The training manual discusses that in order to have a positive relationship with a student, the tutor needs to motivate them to do their best. The learning center's blog shares about the significance of parents and tutors building a relationship with their students through motivation. Lastly, the customer reviews revealed that tutors motivating students brought the students to reach and strive further than their academic goals.

**Research sub-question 3: Innovation**. Research sub-question 3 asked, "How does one successful K-12 learning center for students with learning disabilities in Southern California use innovation to sustain their business?" This question is paired with the literature theme of innovation and data collected in reference to this question was placed in the code group, "innovation." Within the code group of innovation, three codes were discovered: embracing change, neuroscience, and technology integration.

*Embracing change.* Innovation, the creation and implementation of new ideas, cannot fully exist without embracing change. In his interview, the new tutor talked about how "the company is ambitious and has a lot of ideas for the future." In his first company meeting, he said that there was a discussion about the small improvements that immediately affect the tutors, and the larger changes that are planned to take place in the future. In the interview with the operations manager, he discussed change on two levels: changing how students think and the fast-paced, changing world. On a student level, the operations manager believes that "for a student to really become different, become better, he or she has to engage changing their core." He discussed how a student must change the way they think to change and improve their learning habits. On another level, the operations manager also stated that he wants to keep the company current with the changing technology and world around us. In reference to this, he stated:

The problem is, if you don't embrace change, change will eat you alive. The pace at which technology is moving is mind blowing. The pace at which we are changing the world around us. In the face of so much change, we have to embrace it.

*Neuroscience.* One innovative aspect about the learning center is the connections they make to neuroscience. The chief of academics and the operations manager discussed neuroplasticity as the way in which they work to rewire their students' thoughts in reference to motivation and needs. The operations manager stated that the company believes in "changing he fundamental core...the psyche" of students and that "the spirit of the child is more important than raising the grade." The center is working towards hosting more seminars for their tutors in relation to the correlations made between psychology, brain science, and education.

*Technology integration.* The last code of innovation is the most prevalent within the company, integrating technology in their tutoring sessions. The company website mentions

striking a balance between integrating new technology "with time-tested teaching methods" to "contribute to an optimal learning experience." Through observations at the learning center, the researcher discovered that there were televisions and touch screen computers paired with projectors in each classroom, along with use of the internet and academic websites during student sessions. The operations manager discussed the importance of embracing new technology and that "technology allows you to take small resources and make big gains."

**Research sub-question 4: Structure.** Research sub-question 4 asked, "How does one successful K-12 learning center for students with learning disabilities in Southern California use structure to sustain their business?" This question is paired with the literature theme of structure and data collected in reference to this question was placed in the code group, "structure." Within the code group of structure, five codes were discovered: employment, location and space, report system, scheduling, and vision.

*Hiring process.* One category of structure that affects the learning center is the center's hiring process. New employees are found and hired by word-of-mouth or by using an online job search database such as www.indeed.com. As stated by the operations manager, the employment process is "a very rigorous, threefold process." This involves an interview, an assessment test based on the subjects the applicant is interested in teaching, and a recorded, mock teaching session. Employees hired at the center who are full-time have roles in accordance with or separate from tutoring, such as the chief of academics and the receptionist. Those who are hired on only as tutors are part-time employees, ranging from two to fifteen hours per week. The part-time tutors are contractors, or W2 employees. The operations manager explains the tutor jobs as supplemental income and a compliment to their outside schooling and careers. Compensation for

the tutors is suggested based on the compensation of other tutors at the center, and agreed upon by the newly hired tutor and operations manager.

*Location and space.* the learning center is located in a busy, metropolitan area in Southern California in a large office building on the first floor. The customers of the learning center live, work, and go to school in the surrounding metropolitan area. There is no free or lowcost parking available for the learning center – the operations manager said that this does not make a difference to his customers because "if they drive, they are not coming here." The operations manager stated this to mean that students who drive their own vehicles do not drive them to the learning center. Students arrive at the center either by being dropped off, by walking, biking, or taking a bus. The learning center has two entrances, a back entrance and a front entrance. The front entrance leads to the reception area, with the reception desk facing away from the front door. The waiting area for the students is a large study hall classroom where students can have a quiet space to study and complete assignments before and after their tutoring sessions. The learning center has a couple of business offices for management, a couple of large classrooms for meetings, lectures, and group sessions, and a handful of smaller classrooms for one-on-one and small group tutoring sessions. The operations manager noted that the walls in the classrooms are intentionally left blank to help students focus and avoid distractions.

*Report system.* A report system is in place for the tutors to inform management and the parents about the progress of their sessions with the students. Each tutor gives a brief report about each tutoring session. The tutors also report their time spent in sessions through an employee timesheet. This timesheet includes the date, time in and time out, and a description of the service offered. The tutor and management must sign off on the timesheet to declare accuracy of the record.

83

*Scheduling.* The operations manager revealed the significance of scheduling and that the job of the company is solve the problem of time and place to conduct sessions. He went on to say:

Scheduling is the lifeblood of a company. It is also the most difficult aspect of the matching process. The person who figures out this algorithm is the person who's going to become the most efficient [person] in this business.

The learning center allows for "a lot of slack" in their scheduling by keeping many tutors on staff to cater to the needs of the students. Tutors are scheduled to work with students based on their subject area and availability. Availability and scheduling is charted online using the program Outlook. Every member of the company has access to the scheduling calendar. According to the operations manual, when a tutoring session is set on the calendar, the tutor belonging to that session must accept the event before it is confirmed.

*Vision.* The learning center's structure is based on their company vision. The vision, philosophy, goals, and mission statement are all available in the teacher training manual and the operations manual. The website also shares information about the company vision. The learning center's vision involves focusing on specific student needs, providing high quality tutoring services, and having a passion for teaching.

**Research sub-question 5: Customer development.** Research sub-question 5 asked, "How does one successful K-12 learning center for students with learning disabilities in Southern California use customer development to sustain their business?" This question is paired with the literature theme of customer development and data collected in reference to this question was placed in the code group, "customer development." Within the code group of customer development, four codes were discovered: customer reviews, discounts, marketing, student improvements.

*Customer reviews.* The customer reviews, available by searching the company name online, are positive recommendations of the learning center. Several of the reviews state that they highly recommend the learning center and that it is beneficial to students who are struggling academically. With over twenty positive reviews available online, the one negative statement found was in reference to the center being too costly.

*Discounts.* The learning center provides price discounts in various ways. The chief of academics stated that when a student first starts at the center, they are given a large discount on a costly aptitude test. Another factor in discounts is that if a student is not reaching their goals after multiple sessions, the student and family are offered extra class sessions for free to help the student achieve their goals.

*Marketing.* The learning center advertises using sources such as a blog, information available on their website, word-of-mouth, and prevalence on the internet when using a search engine. The operations manual has a section about marketing online and offline, including information about seminars, emails, and social media. The researcher also observed that the company logo is clear and distinctive, making it recognizable to the public when seen various times in different mediums.

*Student improvements.* In the interview with the chief of academics, student improvements were discussed as a way to help develop new customers and retain current customers. Quantitative measures, such as test scores, show how students have improved over time while attending sessions at the learning center. Also, students at the center have "about a

75% to 80% of their first-choice college acceptance rate," which plays an important role in customer acquisition and retention at the learning center.

#### Themes

Through thematic analysis, five key themes inductively emerged from the data collected in response to the overarching research question: What strategies does one successful K-12 learning center for students with learning disabilities in Southern California use to sustain their business? These themes are representative of how the learning center continues to sustain their business. The themes are as follows: Theme 1: Engaging in closed-loop communication with all stakeholders; Theme 2: Taking a holistic approach to student improvement; Theme 3: Providing differentiated instruction for a personalized experience; Theme 4: Engaging in a growth mindset; Theme 5: Setting an intention for the learning center to follow.

Theme 1: Engaging in closed-loop communication with all stakeholders. Closed-loop communication means that a team exchanges information clearly, acknowledges getting the information, and confirms their understanding of the information. This type of communication means that both the sender and the receiver of the information being communicated confirm that they understand what is being asked for or stated (McIntyre & Salas, 1995). The researcher witnessed examples of closed-loop communication at the learning center through varying mediums during her four days of observation time. This type of communication was also used between the operations manager and the researcher when setting up times for data collection at the learning center. Closed-loop communication is used in areas such as feedback and consultation.

As stated previously, instead of using "self-report" of tutors assessing their own performance, tutor feedback is given through conversations with parents and students. Once feedback is given, that feedback is shared between management and the tutor and may result in improvements to tutoring sessions or continued excellence in their teaching methods.

Student progress is also relayed to parents during meetings scheduled throughout a student's time at the center. The chief of academics described how he uses closed-loop communication with parents in relation to student progress as:

If we see improvement or if we don't see improvement, either way, we call in the parents and explain what we did, where we are, and where our goals are and why we are seeking those goals. We make sure the parents understand and confirm understanding before they leave the room.

The operations manual's section on consultations includes information on appropriate way to communicate with clients of the learning center. When consultations take place, meetings last until there is a mutual understanding of the steps moving forward between both the learning center and the client involved.

Theme 2: Taking a holistic approach to student improvement. Setting itself apart from learning centers with a sole purpose of student academics, the learning center places a focus on improving students as a whole, including their motivations, behaviors, and academics. Theme 2 was prevalent in many codes, including teaching motivation, building confidence, and neuroscience.

In relation to teaching motivation, the management and tutors at the learning center focus on the students' behaviors and interests to motivate them academically. Similarly, the management and tutors use what they learn about the students' behavior and interests to help build their confidence. The online reviews show the learning center's success in motivating students. In reviews from students and parents, there is an outpouring of gratitude to the learning center and the tutors for aspects beyond raising grades and test scores. One parent reviewer stated, "This center will help students motivate themselves to achieve beyond their expectations." This quote shows how students can be motivated at the learning center past their own academic expectations. The learning center works towards improving students holistically, not only gaining an academic advantage, but also learning to be comfortable with who they are and what they have to offer, giving them the opportunity to enjoy themselves. A review of the learning center written by a student stated:

[The learning center] was a great experience for me because I met teachers who were so willing to help and ensure the best for me, while being easy to work with and constantly available. Without their help, I wouldn't be enjoying my second semester of senior year.

On the topic of college preparation, a student reviewer shared, "Going to [the learning center] taught me to represent myself in the most honest way possible rather than to conform in what we believe the colleges are looking for."

The belief of the center is to understand students' motivation and needs, not just academically, but also socially in how the students relate to themselves. As the chief of academics stated, "the spirit of the child is more important than raising the grade."

Theme 3: Providing differentiated instruction for a personalized experience. The learning center places an emphasis on differentiated instruction in order to give students a personalized experience. Instead of following the same plan for each student based on their academic subject, the learning center develops a plan for each student based on their current needs and learning styles. This theme is proven through how the learning center teaches about learning styles and their company vision.

The chief of academics has at least a one-hour session with each new parent and student to create what he calls a "personal profile analysis." He stated, "This kind of special care is our principle value of how we train our tutors." The chief of academics went on to describe how the "special care" includes creating plans for each student based on their individual needs.

The learning center also provides a brain aptitude test for each student to better understand their learning styles and academic inclinations. When a tutor begins to work with a new student, the chief of academics meets with them to discuss the student's personal profile and provides them with the knowledge and advice they need to personalize student learning.

At [the learning center], we understand that every student is unique, each with a different set of skills, academic needs, and learning styles. For this reason, we design a learning plan specifically tailored to meet each student's needs.

The reviews of the learning center also identify the center's ability to personalize teaching for each student. On the discussion of college preparation, one student reviewer stated:

[The learning center] is really there to connect you to a range of colleges that not only fits your personal profile but the colleges' profile as well. My experience was helpful by exposing me to colleges in and out of my grasp and by preparing for my interviews. My teachers were very professional and were skilled in leading me based on what I needed.

Theme 4: Engaging in a growth mindset. Having a growth mindset means believing that one can continually grow and adapt through continued hard word and acceptance of change (Dweck, 2017). The learning center is innovative in their thoughts and actions and is continually looking for ways to improve its current state. Having a growth mindset means that the center is open to feedback and continual improvements and does not stay stagnant. The center is not rigid in their methods for running a business or for teaching. They are open and willing to explore

changes based on student outcomes and feedback given. This is proven through embracing change and technology integration.

The learning center embraces change by continually looking to improve their current methods and giving everyone who works at the center a chance to take part in the brainstorming process. The new tutor described a meeting with management and employees during his interview as a "place where change happens." He said:

During the meeting they showed us how they are changing doing time logs and asked what we thought. They also told us where they are going with the company in the future and how changes are made. The company is ambitious and has a lot of ideas for the future. I'm hoping I can help out.

The learning center also embraces change in their methods and their technology. In his interview, the operations manager made references to keeping the center's technology and methods up-to-date with current educational research and technological advances. He stated:

Technology allows you to take small resources and make big gains. In the face of so much technological change, we have to embrace it.

Observations of each classroom brought to light that varying types of technology are used with the students, including televisions, projectors, and computer resources. Tutors also use educational applications and websites that they discover through the internet to find new ways to help their students.

Theme 5: Setting an intention for the learning center to follow. The learning center is very intentional in their actions and does not make decisions or changes without a specific purpose to bring success to both the students and the center itself. This theme is seen through the company's vision and through the use of their location and space.

The vision of the learning center sets strong intentions that hold more meaning and specificity than simply having the goal of improving academics. Part of the vision of the company states:

Our goal is to empower students with the knowledge and study skills necessary to become independent thinkers and learners, helping students gain the confidence they need to find their own paths to success.

The learning center has the goal to give students confidence and independence, which in turn works to boost their academics.

The set-up of the learning center was created with the intention of giving students a variety of spaces to work with very little distraction. There are varying sized classrooms to fit singular student or group learning needs. In each of these rooms, the walls are bare and extra materials including books and school supplies are not present. The learning center purposefully gives students a bare space to help them clear their minds and focus on their current task. The operations manager stated, "The rooms are clear and blank for a reason. There is an intention behind it. Having no distractions in the room really helps the kids focus."

### **Summary of Key Findings and Themes**

Each of the five research sub-questions (training, relationships, innovation, structure, and customer development) acted as code groups during the data analysis process. Each code group was related to a set of codes found inductively during the qualitative data collection using the program, HyperRESEARCH. Within the code group training, it was discovered that tutors are trained at the learning center through rules and discussions regarding expectations and best teaching practices. In regard to relationships, the learning center places an importance on tutors making a connection with their students. Innovation at the learning center is brought about by

embracing technological change. The structure of the learning center follows their vision of focusing on specific student needs, providing high quality tutoring services, and having a passion for teaching. Lastly, customer development plays a role at the learning center by proving success with current and past students based on scores and customer reviews.

The five themes were developed inductively from the data analysis in response to the overarching research question provide clarity of how the learning center continues to sustain their business: The themes are as follows: Theme 1: Engaging in closed-loop communication with all stakeholders; Theme 2: Taking a holistic approach to student improvement; Theme 3: Providing differentiated instruction for a personalized experience; Theme 4: Engaging in a growth mindset; Theme 5: Setting an intention for the learning center to follow.

### **Chapter Five: Conclusion, Implications, and Recommendations**

Chapter Five provides a brief overview of the study including the problem, purpose, methodology, and research questions, as well as a discussion of the key findings and themes discovered from the data analysis. This chapter also provides implications for practice, study conclusions, recommendations for further study, and a final summary.

### **Problem Statement**

Currently, the varied academic needs of students with learning disabilities throughout the U.S. and in Southern California, specifically, has driven demand for private learning centers. When student needs are not fully met within the school system during school hours, teachers recommend and parents seek outside help from learning centers (Park et al., 2016). However, these learning centers often struggle with overall business success and the retention of employees and clients alike. Further, a lack of empirical research has been conducted on learning centers and similar educational businesses directed toward primary and secondary school students. There is limited data about learning centers, what makes them high-performing, and what strategies they use to sustain their current businesses in order to assist students academically. The most recent study depicting the details of a private learning center for K-12 students is a dissertation discussing an assessment tool for the center that was written over two and a half decades ago (Zoll, 1990). Therefore, there is a need and an opportunity to explore strategies to sustain learning centers for students with learning disabilities through the study of one successful learning center.

### Purpose

The purpose of this qualitative case study was to explore and describe strategies used by a successful K-12 learning center for students with learning disabilities in Southern California to

sustain their businesses. This case study took place at one purposefully selected learning center catered to students with learning disabilities in Southern California that was currently successful in their ability to sustain their business. This qualitative case study incorporated data collection through interviews, observations, and archival data (Yin, 2014).

#### **Theoretical Framework (Summary)**

The theoretical frameworks used for this study were a two-part compilation of a business plan approach compiled by a UNESCO team called The Smartest Investment: A Framework for Business Engagement in Education and the Resource Based Theory of Entrepreneurship. The Framework for Business Engagement in Education is a guide for education businesses to create learning opportunities while making a lasting and beneficial business. The Resource Based Theory of Entrepreneurship is a theory with conditions that led to business success, including use of diverse resources, limiting competition, staying innovative, and recognizing entrepreneurship opportunities.

#### **Research Questions**

The overarching research question for this purposefully selected case study was the following:

 What strategies does one successful K-12 learning center for students with learning disabilities in Southern California use to sustain their business?

The sub-questions for this study were:

- A. How does one successful K-12 learning center for students with learning disabilities in Southern California use training to sustain their business?
- B. How does one successful K-12 learning center for students with learning disabilities in Southern California use relationships to sustain their business?

- C. How does one successful K-12 learning center for students with learning disabilities in Southern California use innovation to sustain their business?
- D. How does one successful K-12 learning center for students with learning disabilities in Southern California use structure to sustain their business?
- E. How does one successful K-12 learning center for students with learning disabilities in Southern California use customer development to sustain their business?

### **Discussion of Key Findings**

The key findings of this study include the data found supporting the five research subquestions (training, relationships, innovation, structure, and customer development) and the five inductively developed themes found through the data analysis: Theme 1: Engaging in closedloop communication with all stakeholders; Theme 2: Taking a holistic approach to student improvement; Theme 3: Providing differentiated instruction for a personalized experience; Theme 4: Engaging in a growth mindset; Theme 5: Setting an intention for the learning center to follow.

Summary of the five research sub-questions and code groups. Through qualitative coding, evidence was found in the data collected to answer all five research sub-questions. Code groups were deductively created for each research sub-question, with individual codes assigned to their appropriate sub-group. Within the code group training, it was discovered that tutors are trained at the learning center through rules and discussions regarding expectations and best teaching practices. The learning center trains employees to work specifically with each student and places an importance on motivation and confidence. In regard to relationships, the learning center asks tutors to make a connection with their students about academics and the students' own, personal interests. Innovation at the learning center is brought about by embracing

technological change. The learning center is also open to feedback and suggestions in order to make room for new teaching methods and technology. The structure of the learning center follows their vision of focusing on specific student needs, providing high quality tutoring services, and having a passion for teaching. Lastly, customer development plays a role at the learning center by proving success with current and past students based on scores and customer reviews.

Theme 1: Engaging in closed-loop communication with all stakeholders. Closed-loop communication means that a team exchanges information clearly, acknowledges getting the information, and confirms their understanding of the information (McIntyre & Salas, 1995). The learning center uses clear communication in everything they do because without it, they would not see results. When management uses closed-loop communication with the tutors, the tutors understand what is being asked of them and the steps they need to follow. When tutors communicate with the students, the students have a better understanding of what they are asked to do, which gives them a boost of confidence. When management uses closed-loop communication with clients and has meeting with parents to discuss student progress, the parents become more trusting and accepting of the work the learning center does with their children.

Closed-loop communication relates to the literature discussing the importance of relationships, especially with employees and clients. Relationships form through communication and keeping an open line of communication within a business leads to the innovation of better business practices. Also, communication with educators and parents increases parent awareness and involvement in their child's academics. When parents have a strong relationship with their child's educators, they are more willing to reinforce positive academic behaviors at home (Epstein, 2002). Closed-loop communication is also used during peer observations, a vital

educational component found in the literature and the learning center (Bell & Mladenovic, 2008; Truuvert, 2014). Peer observations are one way that the learning center communicates the most beneficial way to run students' sessions.

Theme 2: Taking a holistic approach to student improvement. By taking a holistic approach, the learning center focuses on the academic and mental state of the students, including behaviors, motivations, and levels of confidence. They do this because they see the students as a whole person that cannot and should not compartmentalize aspects of their life, including their academics. The learning center recognizes that confidence is not only built by solving an academic problem or scoring high on a test, but also by building friendships, relationships, and positive connections with the people around them. Everyone who takes a part in a student's life, especially in a mentorship role, is a great influence to that student. In recognition of this, the learning center has made it a priority for the tutors to connect with their students in a way that builds trust, understanding, and mutual respect.

The notion of a holistic approach was not prevalent in the literature related to small businesses or educational institutions. Epstein (2002) noted the importance of making connections with individuals to build stronger relationships, but did not go into depth about how those connections should be made or if they are made by connecting to students through a level beyond their academics. The importance of a holistic approach in sustaining an educational business is a finding that makes a new contribution to the literature.

Theme 3: Providing differentiated instruction for a personalized experience. Differentiated instruction involves teaching students according to their own learning strategies and styles (Caruana, 2015; Haley et al., 2013). The learning center recognizes that not all students should be treated the same and given the same type of learning path. Understanding the specific learning needs of each student and using individualized strategies has worked for the learning center in the improvement of student academic performance. The chief of academics educates the tutors about each new student's specific needs and how to effectively implement learning strategies for them.

The learning center places an emphasis on differentiated instruction, a method proven to effectively help students with learning disabilities improve academically (Caruana, 2015; Haley et al., 2013; Sadioglu et al., 2013; Stroggilos & Xanthacou, 2006; Zickel & Arnold, 2001). The learning center follows the recommendations of the research in that it understands the importance of teaching students in a variety of ways that best suits their learning needs.

Although differentiation is an effective teaching strategy proven by the literature, it is proven to boost student academics, not proven to run a business. The researcher infers that through the use of differentiated instruction, the learning center sustains itself by achieving positive student outcomes, thus encouraging more customers to become a part of the learning center.

Theme 4: Engaging in a growth mindset. Having a growth mindset means believing that one can grow and adapt through continued hard word and acceptance of change (Dweck, 2017). A growth mindset, being open to feedback, suggestions, and improvement, allows the learning center to incorporate new concepts and practice flexibility of thought. The learning center follows Dweck's recommendations of a growth mindset by believing that both the employees and students at the learning center have the potential to continually expand their knowledge and adapt to change. The learning center chooses to have a growth mindset and keeps an open mind for changes in teaching methods, technology, and adapting to new information that is shared throughout the world every day. When the learning center has a growth mindset and is open to improvements and changes, they allow room for themselves to test out new strategies for success and to see how they may benefit from these new strategies.

This finding interrelates to the literature through the concept of innovation. When innovative opportunities are sought out by small businesses such as the learning center, the business has a competitive edge to stay successful and grow financially (Esterhuizen et al., 2012). Innovation cannot exist without a growth mindset. Innovation in the learning center leaves room for flexibility and creativity. Because the learning center is a small business, they have the ability to experiment with changes in a smaller scale, leaving room to make potential errors and corrections (Bush, 2016).

A growth mindset allows the learning center to be both creative and financially competitive. Being open to new ideas and innovation is having a growth mindset and it is also a component of The Resource Based Theory of Entrepreneurship, one of the theoretical frameworks of this study. Having an open mind to explore new innovative ideas is what gives small businesses, such as the learning center, a competitive advantage over others (Alvarez & Busenitz, 2001).

A growth mindset, through the use of innovation, is also mentioned in the other theoretical framework of this study, The Framework for Business Engagement in Education. One of the first components of how an educational company should "make the business case" is by "fostering innovation in education" (UN Special Envoy for Global Education et al., 2013, p. 12). By incorporating innovation in a company and allowing that innovation to have room to grow, the company is in the practice of a growth mindset.

Theme 5: Setting an intention for the learning center to follow. The learning center is very intentional with many aspects of their company, such as their reasoning for why they teach

how they teach, why their space is structured the way it is, and why they place an emphasis on keeping students motivated. When employees are given a task, they respond more positively when a motivation is given for completing the task.

Intentionality relates to the literature describing the importance of the structure of a business through the use of the symbolic frame as described by Bolman and Deal (2017). In the symbolic frame, it is important to have clear goals and tight coordination through teamwork and understand roles in a business. Employees have the opportunity to truly understand their roles if they understand why they are in those roles and performing their tasks. By having clear, easy to understand, measurable goals, employees are more willing to work towards their goals.

### **Implications for Practice**

Three implications for practice are provided that were born from aspects of the five themes based on the research at the learning center. The three implications for practice are: (a) Have a clear vision statement that differentiates you from other learning centers, (b) Cultivate current clients to help the business succeed and grow, and (c) Focus beyond measurable learning outcomes and take a holistic approach to education.

Have a clear vision statement that differentiates you from other learning centers. Learning centers have the opportunity to differentiate themselves from their competitors by making their vision statement unique to their own business and keeping the intention of that statement a part of everything happening at the center. The intention of the vision statement should be known by every employee and should be a part of their daily tutoring practice. The vision statement should clearly state the reason that the learning center continues to thrive.

A learning center can sustain itself when it has a reason and a purpose for its actions. If the vision of a learning center is simply to give students academic support and raise their grades, then there is not enough specificity for tutors to understand how to carry out the actions of supporting the students. This is because the act of having a goal does not give the tutor instructions on the best way to reach that goal. Without its own unique purpose, a learning center does not gain a competitive advantage, which is beneficial to sustaining a business as theorized by The Resource Based Theory of Entrepreneurship (Alvarez & Busenitz, 2001).

**Cultivate current clients to help the business succeed and grow.** Cultivating current clients of a learning center is more beneficial to the center's growth than searching for new clients. By providing current clients with unparalleled service and getting the students to reach their goals and potential, a learning center can gain success. When current clients are pleased with their time at the center, they are likely to continue learning at the center and recommend it highly to their peers. Positive student results lead to referrals, as demonstrated by the data collected in this study. Referrals are a successful form of advertising that not only helps the center gain more clients, but also improves the current students and environment along the way.

**Focus beyond measurable learning outcomes and take a holistic approach to education.** A learning center should not be solely focused on the academic achievements of their students. By taking a holistic approach, tutors see that students need more than just a passing grade at school or to get into a college. The tutor has the ability to see the student as a person with interests, unique behaviors, and motivations for succeeding not just in school, but also in life skills. Making this type of connection with a student creates trust between the student and the tutor. When students are trusting of their teachers, they become more flexible and open to growth and improvement.

Learning centers can take a holistic approach to education by having management place an emphasis on the importance of tutors getting to know their students and client families. It is important for either tutors or management at the learning center to spend time getting to know both the student and their family or support network. When a tutor is prepared for a new student, they can be given information about the student including specifics on their academics, extracurricular interests, and personality. Tutors can also be placed with students based on shared interests as a way to build trust and a connection between the tutors and the students. This is, as stated by the operations manager, placing a focus on "the spirit of the child."

### Conclusions

Three conclusions can be drawn based on the code groups of inductively discovered codes and the themes found through the data analysis. These conclusions can be applied to the practice of sustaining the business of a K-12 learning center for students with learning disabilities.

Conclusion 1: A learning center can be sustained through a focus on the five literature themes of training, relationships, innovation, structure, and customer development. The five themes found in the literature review, which also represent the five research sub-questions and the five code groups of data analysis, each play an important role in sustaining a learning center business. Training employees through the use of feedback, lectures, orientation, peer observations, teaching learning styles, and teaching motivation give the employees a strong foundation of understanding their roles at the learning center. When relationships at a learning center are built through building confidence, consultation, making connections, and motivation, tutors and students have the opportunity to gain confidence and succeed in their given tasks. Learning centers can stay innovative through embracing change, neuroscience, and technology integration, giving them a competitive edge and keeping them sustained and continually improved by paying attention to employees, having a report system, maintaining scheduling, and having a clear vision. Finally, customers can be developed and maintained through positive customer reviews, discounts, marketing, and continual student improvements.

The theoretical framework from The Smartest Investment: A Framework for Business Engagement in Education, highlights the importance of each of the five literature themes for education businesses as well. Training is a vital component of an education company in order to strengthen the workforce and expand business opportunities for the company and the employees. Building positive relationships in a business helps to develop partnerships and bring the community together. Innovation is a part of making the case for why a business is necessary and that a business should have a clear vision. Having a strong structure of a business will build sustainability and potentially boost wages. Lastly, developing customer relations promotes business growth (UN Special Envoy for Global Education et al., 2013).

The Resource Based Theory of Entrepreneurship, the second theoretical framework that assisted in guiding the research for this study, connected to the literature themes of relationships, innovation, and structure. This theory established the importance of relationships in a business by stating that a business owner looking for a competitive advantage should be familiar with their competitors. Also, by building relationships, a business may gain new ideas and opportunities to benefit business growth. The theory connected largely to the concept of innovation and how the use of innovation limits competition. The structure of a business in this theory refers to the benefit of having a diversity of business resources.

The five themes developed from the research and the coding of the data also relate strongly to the literature themes and how they work to sustain a business. The first theme, engaging in closed-loop communication with all stakeholders, relates to both training and relationships. The learning center trained employees to practice closed-loop communication. The center also built strong relationships with employees and clients through the use of this clear form of communication. The second and third themes, taking a holistic approach to student improvement and providing differentiated instruction for a personalized experience, both relate to innovation. Placing a focus on the student as a whole person, including their academics, interests, behaviors, and motivations, is an innovative concept for a learning center. This type of focus gives students personalized attention and confidence to succeed in life skills as well as academics. The fourth and fifth themes, engaging in a growth mindset and setting an intention for the learning center to follow, relate mainly to the structure of a business. The vision statement of the learning center, a key component of the structure of a business, shows the center's true intention. The vision proved that the center has a growth mindset by acknowledging the growing and changing needs of the students as a focus of the center.

**Conclusion 2: Flexibility allows for individualization and continual improvement.** A successful learning center cannot be rigid in its thoughts or actions. In order to achieve long term success, learning centers need to be open to adaptability, including changes in structure, management strategies, teaching styles, and technology. A willingness to accept growth and change sets a platform for endless possibilities of improvement. Flexibility leads to innovative practices, allowing a business to sustain and grow over time.

Flexibility relates to The Resource Based Theory of Entrepreneurship, as an approach to helping businesses grow, innovate, and improve. According to this theory, being flexible means being open to using diverse resources that help a business limit their competition. When a business is flexible and open to suggestions, it is easier for the business to try new ideas that their competition are not using.

Flexibility also relates to the third theme found through the research and coding of the data, providing differentiated instruction for a personalized experience. This theme emphasizes the importance of recognizing each student as more than the sum of their academic parts. The learning center used differentiated instruction by giving each student a lesson plan that suits their learning styles as opposed to staying rigid and creating one plan to fit all students learning a particular subject. The implementation of differentiated instruction shows the flexibility of an educator. When businesses are flexible, they are open to trying new methods and experiences as opposed to falling into repetitive patterns, giving them the chance to learn, grow, and improve over time. Through a flexible environment and mindset, a learning center can indefinitely find way to continually improve in all aspects of the business.

**Conclusion 3: Learning centers seeking a competitive advantage should focus on innovation.** A successful learning center will continually find ways to be innovative, giving them a competitive advantage. As related to the Resource Based Theory of Entrepreneurship, staying innovative by creating new business strategies to limit imitation from other businesses will lead to business success and create a competitive advantage (Alvarez & Busenitz, 2001). Learning centers can also stay innovative through their use of technology by staying up-to-date using the latest technological devices and ever-expanding resources found through the internet. This gives students a chance to connect with the technological changes taking place in the world around them and prepares them for life outside of the classroom. Innovation also relates to the theme of taking a holistic approach to student improvement. It is a competitive advantage, a strategy that is not commonly used by learning centers, to focus on the student as a whole as opposed to focusing solely on student academics.

### **Recommendations for Further Study**

This study combines the topics of business and education in order to explore and describe strategies used by successful learning centers to sustain their businesses. There is very little research directly related to the problem of the study; therefore, further research is recommended to gain a better understanding of strategies used in learning centers to sustain their businesses. Researchers may consider a variety of study types for further research on this topic that include a multiple case study, a longitudinal study, and a mixed method study to involve qualitative and quantitative data.

**Multiple case study.** As this was a singular case study, one recommendation is to complete a multiple case study to allow for comparison of strategies between two or more learning centers. The learning centers involved in the multiple case study could be chosen from a variety of areas to broaden the scope of the study past Southern California. With the right resources, the multiple case study could also include learning centers from a variety of states across the United States. This would give the researcher a broadened perspective on whether the strategies used by one learning center translate well in other learning centers. The researcher would also be able to determine if there are consistencies in the strategies used between the different learning centers.

**Longitudinal study.** Another recommendation is to complete a longitudinal study. This study collected data over the course of a couple months and was not longitudinal. By collecting data over a longer period of time, such as the course of a year, the researcher may be able to see new and differing developments. Strategies may be seen to adapt and evolve over a longer period

of time. Also, strategies that are seen as successful one month may prove to be less effective over a longer stretch of time.

**Mixed method study.** A final recommendation for further study is to follow a mixed method approach. The combination of qualitative and quantitative research gives the researcher an opportunity to compare perspectives of the data using contrasting methods. Also, qualitative results would have the opportunity to be supported with quantitative data (Creswell, 2014). For a learning center, this may include proving success of the center's teaching methods through the academic improvements of the students based on quantitative test scores.

### **Concluding Thoughts**

Learning centers for K-12 students with learning disabilities are small businesses in education that have the intention of improving student outcomes while also sustaining themselves as a business. There is a great need for these types of learning centers to not only exist, but to also be successful in their continued excellence. This study explored how one successful K-12 learning center for students with learning disabilities continues to thrive and sustain their business.

The theoretical frameworks that acted as an inspiration throughout the literature and the study as a whole were The Framework for Business Engagement in Education and The Resource Based Theory of Entrepreneurship. The Framework for Business Engagement in Education is a guide for education businesses to create learning opportunities for students while identifying the needs of education businesses. The Resource Based Theory of Entrepreneurship provides insight for the use of innovative resources and practices for entrepreneurs of to create business success and was applied to the small business of a learning center.

107

The five sub-research questions acted as a guide throughout the study and the data collection. They were also used as literature themes and the code groups for qualitative analysis of the data: training, relationships, innovation, structure, and customer development. Using these five components as code groups, codes from the data were inductively found and placed into the groups. The data analyzed stemmed from the three methods of data collection including interviews, observations, and archival data from the learning center.

Once the codes were analyzed and broken down, five themes emerged from the data collection: Theme 1: Engaging in closed-loop communication with all stakeholders; Theme 2: Taking a holistic approach to student improvement; Theme 3: Providing differentiated instruction for a personalized experience; Theme 4: Engaging in a growth mindset; Theme 5: Setting an intention for the learning center to follow. The five themes became the core components of the five implications for practice: (a) Practice closed-loop communication with employees and clients, (b) Focus beyond measurable learning outcomes and take a holistic approach to education, (c) Teach using differentiated instruction rather than following a set formula (d) Engage in a growth mindset for continuous improvement, and (e) Set an intention and be purposeful in your actions. The codes, themes, and implications then led to three conclusions of the study. Conclusion 1: Relationships built around communication and motivation instill confidence and improve outcomes. Conclusion 2: Flexibility allows for individualization, and continual improvement. Conclusion 3: Learning centers seeking a competitive advantage should focus on innovation.

Currently, there is very little research connecting education and business, and, prior to this dissertation, there was only one study connecting business and learning centers. That study, depicting the details of a private learning center for K-12 students, was a dissertation discussing

an assessment tool for the center that was written over two and a half decades ago (Zoll, 1990). Now is the time for educators and researchers to conduct further studies on the business of learning centers as a way to assist current and upcoming learning centers in sustaining their businesses. Giving entrepreneurial educators and businesses owners tools, such as research on the business learning centers and effective business models, will help guide them in the positive direction of sustaining their own businesses. As learning centers for students with disabilities are an important and increasing need for K-12 students, I hope to see more research about sustaining these business, along with more research combining the aspects of sustaining a successful education business in order to give students the attention, academics, and confidence that they deserve.

As the researcher and as an educational therapist with my own learning center for students with disabilities, I was eager to explore the connections between the literature on small businesses and education with the actual discoveries made from the data analysis of the study. I expected the strategies for sustaining the learning center to be more generic and centered around advertising, guidelines for employees, and customer development based on the location of the learning center. I was pleasantly surprised to come to the conclusion that aspects such as flexibility, innovation, and intentionality were the most influential findings in relation to the success of the learning center. In opposition to the generic ideas that I had of advertising, rules, and location, these intangible concepts are what make the learning center unique in its practice and what give it a competitive edge, allowing it to stand out among its competition. As learning centers for students with disabilities are an important and increasing need for K-12 students, I hope to see more research about sustaining these business, along with more research combining the aspects of sustaining a successful education business in order to give students the attention, academics, and confidence that they deserve.

#### REFERENCES

- Alvarez, S. A., & Busenitz, L. W. (2001). The entrepreneurship of resource-based theory. Journal of Management, 27(6), 755. doi:10.1177/014920630102700609
- Amos, J. (2008). Dropouts, diplomas, and dollars: U.S. high schools and the nation's economy. Washington, DC: Alliance for Excellent Education.
- Anoka-Ramsey Community College (2007). What is tutoring? *Introduction to Tutoring: Tutor Training*. Retreived from Student Services Department Web page: http://www. anokaramsey.edu/tutor\_training/module1/ intro\_tutoring.cfm
- Aurini, J. D. (2012). Patterns of tight and loose coupling in a competitive marketplace: The case of learning center franchises. *Sociology of Education*, *85*(4), 373-387.
  doi:10.1177/0038040712441375
- Baker, T., Campbell, S., & Ostroff, D. (2016). Independent school leadership: Heads, boards, and strategic thinking. *Peabody Journal of Education (0161956X), 91*(5), 574-587. doi:10.1080/0161956X.2016.1227165
- Barney, J. (1991). Firm resources and sustained competitive advantage. *Journal of Management, 17*(1), 99–120. doi:10.1177/014920639101700108
- Belitski, M., & Heron, K. (2017). Expanding entrepreneurship education ecosystems. *Journal of Management Development*, *36*(2), 163-177. doi:10.1108/JMD-06-2016-0121
- Bell, A., and R. Mladenovic. 2008. The benefits of peer observation of teaching for tutor development. *Higher Education*, 55(6), 735–52. doi:10.1007/s10734-007-9093-1
- Bernard, H. R., Wutich, A., & Ryan, G. W. (2017). *Analyzing qualitative data: Systematic approaches*. Los Angeles, CA: SAGE.

- Birmingham, J., Pechman, E. M., Russell, C. A., & Mielke, M. (2005). Shared features of high-performing after-school programs: A follow-up to the TASC evaluation.
  Washington, DC: Policy Studies Associates.
- Bolman, L. G., & Deal, T. E. (2017). Reframing organizations: Artistry, choice and leadership.
- Bond, N. (2012). Key success factors for effective team-building activities. *AMWA Journal: American Medical Writers Association Journal*, 27(3), 129-131.
- Bush, M. A. (2016). Strategies affecting the sustainability of small businesses (Doctoral dissertation). Available from ProQuest Dissertations & Theses Global, ProQuest Dissertations and Theses A&I: The Humanities and Social Sciences Collection. (1756771028).
- Calderwood, B. J. (2009). Learning center issues, then and now: An interview with frank christ. *Journal of Developmental Education*, 32(3), 24-27.
- Caruana, V. (2015). Accessing the common core standards for students with learning disabilities:
  Strategies for writing standards-based IEP goals. *Preventing School Failure*, *59*(4), 237-243. doi:10.1080/1045988X.2014.924088
- Christ, F. (1971). Systems for learning assistance: Learners, learning facilitators, and learning Centers. In F. L. Christ (Ed.), *Interdisciplinary Aspects of Reading Instruction, IV, Proceedings of the Fourth-Annual Conference of the Western College Reading Association.* Los Angeles, CA (pp. 32-41).
- Cornman, S. Q., Zhou, L., Howell, M.R., & Young, J. (2017). Revenues and expenditures for public elementary and secondary education: School year 2014-15 (Fiscal year 2015): First look. U.S. Department of Education. Washington, DC: National Center for Education Statistics.

- Cornucopia of Disability Information (2007). Data on students with disabilities. Retrieved December 2007 from http://codi.buffalo.edu/ archives/colleges/stats.htm
- Cortiella, C. & Horowitz, S. H. (2014) *The state of learning disabilities: Facts, trends and emerging issues.* New York, NY: National Center for Learning Disabilities.

Coulter, M. (2013). Strategic management in action. Upper Saddle River, NJ: Pearson.

- Covey, S. M., & Merrill, R. R. (2006). *The speed of trust: The one thing that changes everything*. New York, NY: Simon & Schuster.
- Creswell, J. W. (1998). *Qualitative inquiry and research design: Choosing among five traditions*. Thousand Oaks, CA: Sage Publications.
- Creswell, J. W. (2014). *Research design: qualitative, quantitative, and mixed methods approaches*. Los Angeles: Sage.
- Cunningham, M., Huchting, K. K., Fogarty, D., & Graf, V. (2017). Providing access for students with moderate disabilities: An evaluation of a professional development program at a catholic elementary school. *Journal of Catholic Education, 21*(1), 138-170. doi:10.15365/joce.2101072017
- Dweck, C. S. (2017). *Mindset*. London, England: Robinson, an imprint of Little, Brown Book Group.
- Epstein, J. L. (2002). *School, family, and community partnerships: Your handbook for action*. Thousand Oaks, Ca.: Corwin Press.
- Esterhuizen, D., Schutte, C., & Du Toit, A. (2012). A knowledge management framework to grow innovation capability maturity. *South African Journal of Information Management, 14*(1), 1-10. doi:10.4102/sajim.v14i1.495

- Ferdowsian, M. C. (2016). Total business excellence: A new management model for operationalizing excellence. *International Journal of Quality & Reliability Management*, 33(7), 942-984. doi:10.1108/IJQRM-08-2014-0109
- Fiet, J. O. (2001). The pedagogical side of entrepreneurship theory. *Journal of Business Venturing*, *16*(2), 101.

Fowler, F. (2008). Survey research methods (4th ed.). Thousand Oaks, CA: Sage.

- Gerhardt, S., Hazen, S., & Lewis, S. (2014). Small business marketing strategy based on Mcdonald's. *Asbbs Ejournal, 10*(1), 104-112.
- Gostick, A.R., & Elton, C. (2012). *All in: How the best managers create a culture of belief and drive big results.* New York, NY: Free Press.
- Haager, D., & Vaughn, S. (2013). The common core state standards and reading: Interpretations and implications for elementary students with learning disabilities. *Learning Disabilities Research & Practice*, 28(1), 5-16. doi:10.1111/ldrp.12000
- Haley, M., Hammond, H., Ingalls, L., & Marín, M. R. (2013). Parental reactions to the special education individual education program process: Looking through the lens of grief. *Improving Schools, 16*(3), 232-243. doi:10.1177/1365480213505180
- Holliday, T., (2012). Evaluating the effectiveness of tutoring: An easier way. *Learning* Assistance Review (TLAR), 17(2), 21-32.
- Hord, S. M. (1987). *Taking charge of change*. Alexandria, VA: Association for Supervision and Curriculum Development.
- Hsiu-Li Chen. (2005). A competence-based strategic management model factoring in key success factors and benchmarking. Benchmarking: *An International Journal*, *12*(4), 364-382. doi:10.1108/14635770510609033

- Huang, D., & Cho, J. (2010). Using professional development to enhance staff retention. *Afterschool Matters*, (12), 9-16.
- Jochims, T. (2016). Social reciprocity as a critical success factor for small and mid-size enterprises: Work relationships as reflections of social exchange structures. *Management Revue*, 27(3), 188-207. doi:10.1688/mrev-2016-Jochims
- Kellermanns, F., Walter, J., Crook, T. R., Kemmerer, B., & Narayanan, V. (2016). The resource-based view in entrepreneurship: A content-analytical comparison of researchers' and entrepreneurs' views. *Journal of Small Business Management*, 54(1), 26-48. doi:10.1111/jsbm.12126
- King-Sears, M. (2007). Designing and delivering learning center instruction. *Intervention in School and Clinic*, *42*(3), 137-147. doi:10.1177/10534512070420030201
- Lencioni, P. (2012). *The advantage: Why organizational health trumps everything else in business*. San Francisco, CA: Jossey-Bass.
- Levine, M., & Levine, A. G. (2014). Follow the money: There's no business like the ed. business. *American Journal of Orthopsychiatry*, *84*(4), 377-386. doi:10.1037/h0099848
- Lunenburg, F. C., & Irby, B. J. (2008). Writing a successful thesis or dissertation: tips and strategies for students in the social and behavioral sciences. Thousand Oaks, CA: Corwin Press.
- McIntyre, R.M. & Salas E. (1995). *Measuring and managing for team performance: emerging principles from complex environments*. San Francisco, CA: Jossey-Bass.
- Merlone, L. & Moran, D. (2008). Transition works: Self-awareness and self-advocacy skills for students in the elementary learning center. *TEACHING Exceptional Children Plus, 4*(4)

- Meyer, J. W. & Rowan, B. 1978. *The Structure of Educational Organizations*. San Francisco, CA: Jossey-Bass.
- National Joint Committee on Learning Disabilities (2011). *Learning Disability Quarterly*, *34*(1), 3-16. doi: 10.1177/073194871103400101
- Norton, J. (2002). Job stress among learning center managers. *Learning Assistance Review*, 7(1), 22-36.
- Oana, I. (2012). Small business management strategies. Ovidius University Annals, Series Economic Sciences, 12(1), 540-545.
- Omri, A., Frikha, M. A., & Bouraoui, M. A. (2015). An empirical investigation of factors affecting small business success. *Journal of Management Development*, 34(9), 1073-1093. doi:10.1108/JMD-07-2013-0088
- Paluta, L. M., Lower, L., Anderson-Butcher, D., Gibson, A., & Iachini, A. L. (2016). Examining the quality of 21st century community learning center after-school programs: Current practices and their relationship to outcomes. *Children & Schools, 38*(1), 49-56. doi:10.1093/cs/cdv040
- Park, H., Buchmann, C., Choi, J., & Merry, J. J. (2016). Learning beyond the school walls: Trends and implications. *Annual Review of Sociology*, 42, 231-252. doi:10.1146/annurevsoc-081715-074341
- Payne, E. M., Hodges, R., & Hernandez, E. P. (2017). Changing demographics and needs assessment for learning centers in the 21st century. *Learning Assistance Review (TLAR)*, 22(1), 21-36.
- Penrose, E. T. (1959). The theory of the growth of the firm. Oxford, GB: Blackwell.

- Petersen, J. L. (2014). For education entrepreneurs, innovation yields high returns. *Education Next*, *14*(2), 8-16.
- Philip, M. (2011). Factors affecting business success of small & medium enterprises (SMEs). *Amity Global Business Review*, 6(1), 118-136.
- Radecki, J. (1984). Educational therapy defined. *Journal of Learning Disabilities*, *17*(1), 7. doi:10.1177/002221948401700103
- Radford, J., Bosanquet, P., Webster, R., & Blatchford, P. (2015). Scaffolding learning for independence: Clarifying teacher and teaching assistant roles for children with special educational needs. *Learning and Instruction*, *36*, 1-10.
  doi:10.1016/j.learninstruc.2014.10.005
- Raley, R., Grossman, J., & Walker, K. E. (2005). *Getting it right: Strategies for after-school success*. Philadelphia, PA: Public/Private Ventures.
- Rese, A., & Baier, D. (2011). Success factors for innovation management in networks of small and medium enterprises. *R&D Management*, *41*(2), 138-155. doi:10.1111/j.1467-9310.2010.00620.x
- Sadioglu, O., Bilgin, A., Batu, S., & Oksal, A. (2013). Problems, expectations, and suggestions of elementary teachers regarding inclusion. *Educational Sciences: Theory and Practice*, 13(3), 1760-1765.
- Sánchez, J. C. (2013). The impact of an entrepreneurship education program on entrepreneurial competencies and intention. *Journal of Small Business Management*, 51(3), 447-465. doi:10.1111/jsbm.12025
- Schumann, D. W., Peters, J., & Olsen, T. (2013). Cocreating value in teaching and learning centers. *New Directions for Teaching and Learning*, (133), 21-32. doi:10.1002/tl.20043

- Segool, N. K., Carlson, J. S., Goforth, A. N., von der Embse, N., & Barterian, J. A. (2013).
  Heightened test anxiety among young children: Elementary school students' anxious responses to high-stakes testing. *Psychology in the Schools, 50*(5), 489-499.
  doi:10.1002/pits.2168
- Simpson, M., Tuck, N., & Bellamy, S. (2004). Small business success factors: The role of education and training. *Education + Training*, 46(8-9), 481-491. doi:10.1108/00400910410569605
- Solesvik, M., Westhead, P., & Matlay, H. (2014). Cultural factors and entrepreneurial intention:
  The role of entrepreneurship education. *Education & Training*, 56(8-9), 680-696.
  doi:10.1108/ET-07-2014-0075
- St-Jean, E. (2012). Mentoring as professional development for novice entrepreneurs: Maximizing the learning. *International Journal of Training and Development*, 16(3), 200-216. doi:10.1111/j.1468-2419.2012.00404.x
- Stahl, N. A., & Henk, W. A. (1993). Helping parents to select and evaluate private Reading/Learning centers. *Reading Horizons*, 34(1), 62-70.
- Stroggilos, V., & Xanthacou, Y. (2006). Collaborative IEPs for the education of pupils with profound and multiple learning difficulties. *European Journal of Special Needs Education, 21*(3), 339-349. doi:10.1080/08856250600810872
- Trammell, J. (2005). Learning about the learning center: Program evaluation for learning assistance programs. *The Learning Assistance Review*, *10*(2), 31-40.
- Truschel, J., & Reedy, D. L. (2009). National survey--what is a learning center in the 21st century? *Learning Assistance Review*, *14*(1), 9-22.

Truuvert, T. (2014). Enhancing tutorial learning experiences: A programme to develop sessional-tutor teaching skills by raising awareness about learning. *Studies in Higher Education, 39*(1), 20-33. doi:10.1080/03075079.2012.690731

- UN Special Envoy for Global Education, UNICEF, United Nations Educational, Scientific and Cultural Organization, & United Nations Global Compact. (2013). The Smartest Investment: A Framework for Business Engagement in Education. Retrieved from https://www.unglobalcompact.org/library/391
- Vagg, P. R., & Spielberger, C. D. (1999). The Job Stress Survey: Assessing perceived severity and frequency of occurrence of generic sources of stress in the workplace. Journal of *Occupational Health Psychology*, 4(3), 288-292. doi:10.1037//1076-8998.4.3.288
- Valkenburg, J. (2010). Joining the conversation: Scaffolding and tutoring mathematics. *Learning* Assistance Review, 15(2), 33-41.
- Walker, E., & Brown, A. (2004). What success factors are important to small business owners? *International Small Business Journal*, 22(6), 577-594. doi:10.1177/0126624260404711
- Yang, X. X. (2016). Key success factors of small business in a southern region of california (Doctoral dissertation). Available from ProQuest Dissertations & Theses Global, ProQuest Dissertations and Theses A&I: The Humanities and Social Sciences Collection. (1725903656).
- Yin, R. K. (2014). *Case study research: Design and methods*. London, England: Sage Publication.
- Zapalska, A. M., Brozik, D., & Zieser, N. (2015). Factors affecting success of small business enterprises in the polish tourism industry. *Tourism: An International Interdisciplinary Journal, 63*(3), 365-381.

Zickel, J. P., & Arnold, E. (2001). Putting the I in the IEP. Educational Leadership, 59(3), 71-73.

Zoll, J. G. (1990). A learning center assessment tool for the assessment of academic services of a private learning center. Available from ProQuest Dissertations & Theses Global,
 ProQuest Dissertations and Theses A&I: The Humanities and Social Sciences Collection. (9030767)

### APPENDIX A

### Recruitment Form

Dear [Name],

My name is Rebecca Cohen, and I am a doctoral student at Pepperdine University Graduate School of Education and Psychology. I am conducting a research study to explore strategies used by successful K-12 learning centers for students with learning disabilities in Southern California to sustain their businesses, and you are invited to participate in the study.

This study has the potential assist learning centers, especially that of your own, in achieving business and academic goals through exploring best practices and strategies to sustain your business. The findings of this study have the potential to benefit your business specifically through the data collected and shared by exploring your learning center.

If you agree, you are invited to participate by sharing archival business data, allowing the researcher to observe your work at your learning center, and participating in an interview. The archival data collection, observation, and interview may take a minimum of two visits during the day at your learning center.

Participation in this study is voluntary. Your identity as a participant, as well as the identity of your learning center, will remain confidential during and after the study. Confidentiality will be provided through pseudonyms for participants and a fictitious business name for your learning center. Your personal information and data will be stored on a password protected computer in the principal investigators place of residence, and will only be made available to the research team members.

If you have questions or would like to participate, please contact me at rebecca.klein@pepperdine.edu.

Thank you for your participation,

Rebecca Cohen Pepperdine University Graduate School of Education and Psychology Doctoral Student

### APPENDIX B

## Permission to Conduct Study

To Whom It May Concern:

We would like to proudly open our doors and invite researcher Rebecca Cohen, M.Ed. to study how we help, tutor and mentor students with learning disabilities at our learning centers.

Please forward any relevant information and further communications regarding this matter to XXXXX our branch manager.

Email: XXXXX Phone: XXXXX

Best Regards,

XXXXX Director XXXXX Email: XXXXX Phone: XXXXX ,

### APPENDIX C

### Informed Consent

Your learning center is invited to participate in a research study conducted by Rebecca Cohen, M.A.Ed. and Dr. Paula Thompson at Pepperdine University. Your participation is voluntary. You should read the information below, and ask questions about anything that you do not understand, before deciding whether to participate. Please take as much time as you need to read the consent form. You may discuss the option of participation with family and friends as needed.

The purpose of this study is to explore strategies used by a successful K-12 learning center for students with learning disabilities in Southern California to sustain their businesses.

Your learning center will be asked to share archival business data, allow the researcher to observe work at the center, and participate in interviews. You understand that photos will be taken of the learning center and your interview will be voice recorded if you participate in the interview process. The photos and voice recordings will be used for research purposes only. The data collected, including photos and interviews, will be stored on a password protected computer in the principal investigators place of residence. The data collected will be coded and transcribed.

The potential risks associated with participation in this study include fatigue, boredom, and discomfort sharing information related to your work environment. In the event that you experience any of these risks, you will be given the option of a break. During the interview, you have the right to refuse to answer any question. You understand that if you are not interested in completing your participation in the study, you may discontinue participation at any time. You understand that there is no direct benefit from participation in this study.

I will keep your records for this study confidential as far as permitted by law. However, if I am required to do so by law, I may be required to disclose information collected about you. Examples of the types of issues that would require me to break confidentiality are if you tell me about instances of child abuse and elder abuse. Pepperdine's University's Human Subjects Protection Program (HSPP) may also access the data collected. The HSPP occasionally reviews and monitors research studies to protect the rights and welfare of research subjects.

Your participation is voluntary. Your refusal to participate will involve no penalty or loss of benefits to which you are otherwise entitled. You may withdraw your consent at any time and discontinue participation without penalty. You are not waiving any legal claims, rights or remedies because of your participation in this research study.

If you are injured as a direct result of research procedures you will receive medical treatment; however, you or your insurance will be responsible for the cost. Pepperdine University does not provide any monetary compensation for injury.

If you have any questions regarding the study procedures, please contact Rebecca Cohen at rebecca.klein@pepperdine.edu. If you have further questions, please contact Dr. Paula Thompson, dissertation chairperson of this study, at paula.thompson@pepperdine.edu. If you have questions about your rights as a research participant please contact Dr. Judy Ho, Chairperson of the Graduate & Professional Schools Institutional Review Board at Pepperdine University, 6100 Center Drive Suite 500, Los Angeles, CA 90045, 310-568-5753 or gpsirb@pepperdine.edu.

### Pepperdine IRB Approval



Pepperdine University 24255 Pacific Coast Highway Malibu, CA 90263 TEL: 310-506-4000

#### NOTICE OF APPROVAL FOR HUMAN RESEARCH

Date: December 11, 2017

Protocol Investigator Name: Rebecca Klein

Protocol #: 17-10-624

Project Title: A Case Study of K-12 Learning Centers in Southern California: Exploring strategies to sustain and expand learning centers for students with learning disabilities

School: Graduate School of Education and Psychology

Dear Rebecca Klein:

Thank you for submitting your application for exempt review to Pepperdine University's Institutional Review Board (IRB). We appreciate the work you have done on your proposal. The IRB has reviewed your submitted IRB application and all ancillary materials. Upon review, the IRB has determined that the above entitled project meets the requirements for exemption under the federal regulations 45 CFR 46.101 that govern the protections of human subjects.

Your research must be conducted according to the proposal that was submitted to the IRB. If changes to the approved protocol occur, a revised protocol must be reviewed and approved by the IRB before implementation. For any proposed changes in your research protocol, please submit an amendment to the IRB. Since your study falls under exemption, there is no requirement for continuing IRB review of your project. Please be aware that changes to your protocol may prevent the research from qualifying for exemption from 45 CFR 46.101 and require submission of a new IRB application or other materials to the IRB.

A goal of the IRB is to prevent negative occurrences during any research study. However, despite the best intent, unforeseen circumstances or events may arise during the research. If an unexpected situation or adverse event happens during your investigation, please notify the IRB as soon as possible. We will ask for a complete written explanation of the event and your written response. Other actions also may be required depending on the nature of the event. Details regarding the timeframe in which adverse events must be reported to the IRB and documenting the adverse event can be found in the *Pepperdine University Protection of Human Participants in Research: Policies and Procedures Manual* at community.pepperdine.edu/irb.

Please refer to the protocol number denoted above in all communication or correspondence related to your application and this approval. Should you have additional questions or require clarification of the contents of this letter, please contact the IRB Office. On behalf of the IRB, I wish you success in this scholarly pursuit.

Sincerely,

Judy Ho, Ph.D., IRB Chair

## APPENDIX E

# Interview Protocol and Questions for Owners/Managers

The purpose of this study is to explore strategies used by a successful K-12 learning center for students with learning disabilities in Southern California to sustain their businesses.

## Interview protocol are as follows:

- 1. Introductions between the researcher and the participant.
- 2. Remind participant that they will be audio recorded during the interview.
- 3. Proceed with interview questions.
- 4. Thank the interviewee for their participation.

## Interview question guide for management/owners:

- 1. What is your role at [learning center]?
  - a. How long have you worked here?
  - b. How many employees work here?
- 2. How did you create the learning center? (If applicable)
  - a. Why did you decide to open a learning center?
  - b. How did you decide on a name and branding?
  - c. How did you go about choosing a location?
  - d. What were the greatest struggles and frustrations when starting up?
- 3. What is your approach for hiring and training employees?
  - a. Where do you find new employees to hire?
  - b. How do you decide who to hire?
  - c. What's your onboarding process like?
  - d. How do you determine compensation structure?
  - e. What strategies, if any, do you employ to help employees perform better? (i.e. professional development)
  - f. How do you handle turnover?
  - g. How do you determine how well an employee is performing?
  - h. What are the biggest challenges you've faced with respect to employees?
- 4. How do you manage relationships with customers?
  - a. Relationships with the community?
  - b. Strategies for staying connected with families?
  - c. How do you keep clients and families happy?
  - d. How do you communicate with clients and their family?
  - e. Relationships with other businesses?
- 5. What role does innovation play in the operation and growth of your learning center?
  - a. Do you use technology?
    - i. How do you go about deciding on a technology tool to use and how do you implement it?
  - b. How do you go about measuring and/or improving:
    - i. Client acquisition
    - ii. Client success

- iii. Client retention
- iv. Billing
- v. Employee acquisition
- vi. Employee performance
- vii. Employee retention
- viii. Operations / Efficiencies
- 6. What are the rules and protocols that you have in place for your employees to follow?
  - a. Do you have regularly scheduled meeting with your employees? If so, how often are these meetings and what do they include?
- 7. How do you reach new customers?
  - a. How did you reach your initial customers?
  - b. How do you reach new customers now?
  - c. What kinds of marketing do you do?
  - d. What types of messages appeal most to new customers?
  - e. What has been the most difficult part of acquiring new customers?
  - f. What are strategies and challenges with client retention?
  - g. What are the biggest challenges or frustrations with dealing with clients and their families?
- 8. Do you measure client success and what role does that play?
  - a. How do you communicate progress to clients and their families?

## APPENDIX F

## Interview Protocol and Questions for Employees

The purpose of this study is to explore strategies used by a successful K-12 learning center for students with learning disabilities in Southern California to sustain their businesses.

## Interview protocol are as follows:

- 1. Introductions between the researcher and the participant.
- 2. Remind participant that they will be audio recorded during the interview.
- 3. Proceed with interview questions.
- 4. Thank the interviewee for their participation.

## Interview question guide for employees:

- 1. What is your role at [learning center]?
  - a. How long have you worked here?
  - b. How many employees work here?
- 2. Do you participate in professional development or training for work?
  - a. If so, what does the training entail?
- 3. How do you manage relationships with customers?
  - a. Relationships with the community?
  - b. Strategies for staying connected with families?
  - c. How do you keep clients and families happy?
  - d. How do you communicate with clients and their family?
  - e. Relationships with other businesses?
- 4. What role does innovation play in the operation and growth of the learning center?
  - a. Do you use technology?
    - i. How does the center go about deciding on a technology tool to use and how do you implement it?
  - b. How do you go about measuring and/or improving:
    - i. Client success
    - ii. Client retention
- 5. What are the rules and protocols in place at the learning center?
  - a. Do you attend regularly scheduled meetings with other employees and management? If so, how often are these meetings and what do they include?
- 6. What steps do you take, if any, to ensure customer satisfaction?
- 7. Do you measure client success and what role does that play?
  - a. How do you communicate progress to clients and their families?

# APPENDIX G

# Archival Data Collection Guide

Name of learning center:

Data Type	Data Collected?	Location of Data	Notes
Learning center website	Yes / No		
Information available on the internet (other than website)	Yes / No		
On-boarding	Yes / No		
Professional development	Yes / No		
Types of technology used	Yes / No		
Flyers/brochures	Yes / No		
Other Describe type:	Yes / No		